TECHNICAL REPORTS SERIES NO.

Measurement and Calculation of Radon Releases from NORM Residues

Y. Ishimori, K. Lange, P. Martin, Y.S. Mayya, M. Phaneuf



MEASUREMENT AND CALCULATION OF RADON RELEASES FROM NORM RESIDUES

The following States are Members of the International Atomic Energy Agency:

AFGHANISTAN ALBANIA ALGERIA ANGOLA ARGENTINA ARMENIA AUSTRALIA AUSTRIA AZERBAIJAN BAHRAIN BANGLADESH BELARUS BELGIUM BELIZE BENIN BOLIVIA BOSNIA AND HERZEGOVINA BOTSWANA BRAZIL BUI GARIA BURKINA FASO BURUNDI CAMBODIA CAMEROON CANADA CENTRAL AFRICAN REPUBLIC CHAD CHILE CHINA COLOMBIA CONGO COSTA RICA CÔTE D'IVOIRE CROATIA CUBA CYPRUS CZECH REPUBLIC DEMOCRATIC REPUBLIC OF THE CONGO DENMARK DOMINICA DOMINICAN REPUBLIC ECUADOR EGYPT EL SALVADOR ERITREA **ESTONIA** ETHIOPIA FUI FINLAND FRANCE GABON GEORGIA GERMANY GHANA GREECE

GUATEMALA HAITI HOLY SEE HONDURAS HUNGARY **ICELAND** INDIA INDONESIA IRAN, ISLAMIC REPUBLIC OF IRAO IRELAND ISRAEL ITALY JAMAICA JAPAN JORDAN KAZAKHSTAN KENYA KOREA, REPUBLIC OF KUWAIT KYRGYZSTAN LAO PEOPLE'S DEMOCRATIC REPUBLIC LATVIA LEBANON LESOTHO LIBERIA LIBYA LIECHTENSTEIN LITHUANIA LUXEMBOURG MADAGASCAR MALAWI MALAYSIA MALI MALTA MARSHALL ISLANDS MAURITANIA MAURITIUS MEXICO MONACO MONGOLIA MONTENEGRO MOROCCO MOZAMBIOUE MYANMAR NAMIBIA NEPAL NETHERLANDS NEW ZEALAND NICARAGUA NIGER NIGERIA NORWAY OMAN PAKISTAN PALAU

PANAMA PAPUA NEW GUINEA PARAGUAY PERU PHILIPPINES POLAND PORTUGAL OATAR REPUBLIC OF MOLDOVA ROMANIA RUSSIAN FEDERATION RWANDA SAUDI ARABIA SENEGAL SERBIA SEYCHELLES SIERRA LEONE SINGAPORE SLOVAKIA **SLOVENIA** SOUTH AFRICA SPAIN SRI LANKA SUDAN SWAZILAND SWEDEN SWITZERLAND SYRIAN ARAB REPUBLIC TAJIKISTAN THAILAND THE FORMER YUGOSLAV REPUBLIC OF MACEDONIA TOGO TRINIDAD AND TOBAGO TUNISIA TURKEY UGANDA UKRAINE UNITED ARAB EMIRATES UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND UNITED REPUBLIC OF TANZANIA UNITED STATES OF AMERICA URUGUAY UZBEKISTAN VENEZUELA VIETNAM YEMEN ZAMBIA ZIMBABWE

The Agency's Statute was approved on 23 October 1956 by the Conference on the Statute of the IAEA held at United Nations Headquarters, New York; it entered into force on 29 July 1957. The Headquarters of the Agency are situated in Vienna. Its principal objective is "to accelerate and enlarge the contribution of atomic energy to peace, health and prosperity throughout the world".

TECHNICAL REPORTS SERIES No. 474

MEASUREMENT AND CALCULATION OF RADON RELEASES FROM NORM RESIDUES

Y. ISHIMORI, K. LANGE, P. MARTIN, Y.S. MAYYA, M. PHANEUF

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA, 2013

COPYRIGHT NOTICE

All IAEA scientific and technical publications are protected by the terms of the Universal Copyright Convention as adopted in 1952 (Berne) and as revised in 1972 (Paris). The copyright has since been extended by the World Intellectual Property Organization (Geneva) to include electronic and virtual intellectual property. Permission to use whole or parts of texts contained in IAEA publications in printed or electronic form must be obtained and is usually subject to royalty agreements. Proposals for non-commercial reproductions and translations are welcomed and considered on a case-by-case basis. Enquiries should be addressed to the IAEA Publishing Section at:

Marketing and Sales Unit, Publishing Section International Atomic Energy Agency Vienna International Centre PO Box 100 1400 Vienna, Austria fax: +43 1 2600 29302 tel.: +43 1 2600 22417 email: sales.publications@iaea.org http://www.iaea.org/books

© IAEA, 2013

Printed by the IAEA in Austria September 2013 STI/DOC/010/474

IAEA Library Cataloguing in Publication Data

Measurement and calculation of radon releases from NORM residues. — Vienna : International Atomic Energy Agency, 2013.
p. ; 24 cm. — (Technical reports series, ISSN 0074–1914 ; no. 474) STI/DOC/010/474
ISBN 978–92–0–142610–9
Includes bibliographical references.

1. Radioactive substances — Safety measures. 2. Radioactive wastes — Management. 3. Radon — Measurement. I. International Atomic Energy Agency. II. Series : Technical reports series (International Atomic Energy Agency) ; 474.

IAEAL

13-00835

FOREWORD

The mining and milling of uranium ore produce large quantities of residues containing natural decay series radionuclides. Although such residues are relatively small in magnitude compared with those from metal mining and extraction processes, their present worldwide production exceeds several million tonnes annually. In addition, a number of other resource extraction industries (such as phosphate, mineral sands, oil and gas, and coal) produce residues with similar radionuclide profiles. In recent years, these materials have become increasingly interesting from the point of view of radiological impact assessment. There is thus a need to reduce the environmental and health risks from these materials to an acceptable level.

In response to the needs of its Member States, the IAEA has for many years supported efforts to publish information on the environmental behaviour of radionuclides. In 1992, the IAEA published Technical Reports Series No. 333 (TRS 333), Measurement and Calculation of Radon Releases from Uranium Mill Tailings, which reviewed the major aspects of radon release, control and monitoring as they relate to the management of uranium mill tailings.

Since the publication of TRS 333, a number of developments have taken place in techniques for the measurement, modelling and prediction of radon exhalation, and practical experience has been gained with various residue repositories. It was therefore considered timely to produce a revision of TRS 333 and to expand its scope to include other NORM residues. The present report provides a comprehensive overview of the prediction, measurement and monitoring of radon releases from NORM residues, including uranium mining and milling residues.

The IAEA wishes to express its gratitude to P. Martin (Australia) for his assistance in chairing the meetings, and to K. Lange (Canada) for her assistance in editing this report, as well as to those experts who contributed to the development and completion of this report.

The IAEA officer responsible for this report was M. Phaneuf of the IAEA Environment Laboratories.

EDITORIAL NOTE

This report does not address questions of responsibility, legal or otherwise, for acts or omissions on the part of any person. Although great care has been taken to maintain the accuracy of information contained in this publication, neither the IAEA nor its Member States assume any responsibility for consequences which may arise from its use.

The use of particular designations of countries or territories does not imply any judgement by the publisher, the IAEA, as to the legal status of such countries or territories, of their authorities and institutions or of the delimitation of their boundaries.

The mention of names of specific companies or products (whether or not indicated as registered) does not imply any intention to infringe proprietary rights, nor should it be construed as an endorsement or recommendation on the part of the IAEA.

The authors are responsible for having obtained the necessary permission for the IAEA to reproduce, translate or use material from sources already protected by copyrights. Material prepared by authors who are in contractual relation with governments is copyrighted by the IAEA, as publisher, only to the extent permitted by the appropriate national regulations.

This publication has been prepared from the original material as submitted by the authors. The views expressed do not necessarily reflect those of the IAEA, the governments of the nominating Member States or the nominating organizations.

The IAEA has no responsibility for the persistence or accuracy of URLs for external or third party Internet web sites referred to in this book and does not guarantee that any content on such web sites is, or will remain, accurate or appropriate.

CONTENTS

1.	INT	RODUCTION	1
	1.1. 1.2. 1.3.	Background	1 2 3
2.	FAC	TORS CONTROLLING RADON RELEASES	3
	2.1.2.2.2.3.2.4.	GeneralEmanation coefficient.2.2.1. Factors affecting radon emanation2.2.2. Radon emanation coefficients for natural materialsDiffusion coefficient.Radon exhalation	3 7 9 11 12 16
3.	MEA RAI	ASUREMENT OF VARIABLES AFFECTING DON EXHALATION FLUX	16
	3.1. 3.2.	Measurement of the emanation coefficient.	16 20
4.	EST RAI	IMATION OF VARIABLES AFFECTING DON EXHALATION FLUX	24
	4.1. 4.2. 4.3.	Radon emanation coefficient (E) Diffusion coefficient (D_r) Radium activity concentration (R)	24 24 25
5.	MEA ANI	ASUREMENT OF RADON CONCENTRATION D EXHALATION FLUX DENSITY	25
	5.1. 5.2.	GeneralMeasurement of radon concentration5.2.1. Mode of measurement5.2.2. Nuclear track detector5.2.3. Solid surface barrier detectors5.2.4. Scintillation cell5.2.5. Electret	25 26 26 28 28 28 29 29

		5.2.6. Activated charcoal	30
		5.2.7. Ionization chamber	30
	5.3.	Measurement of exhalation flux density	30
		5.3.1. Accumulation	30
		5.3.2. Flow through method	33
		5.3.3. Adsorption	35
		5.3.4. Measurement of soil gas concentration	36
		5.3.5. Measurement of mass exhalation rate.	37
	5.4.	Calibration and uncertainties	38
6.	SIM	PLIFIED EXPRESSIONS FOR ESTIMATING	
	RAI	OON EXHALATION FLUX DENSITY	39
	6.1.	General	39
	6.2.	Radon exhalation flux density from bare residue	40
		6.2.1. Residue thickness much greater than radon	
		diffusion length	41
		6.2.2. Residue thickness comparable to or less than	
		the diffusion length	41
	6.3.	Radon flux density from residue repositories	
		with a single cover	43
	6.4.	Radon flux density from residue repositories	
		with multiple covers	44
7.	EFF	ECTS OF METEOROLOGICAL CONDITIONS	
	ANI	O THE REPOSITORY COVER	45
	7.1.	Meteorological effects	45
	7.2.	Aspects of repository cover design.	47
		7.2.1. Surface cover effects.	47
		7.2.2. Surface cover materials	48
		7.2.3. Effects of vegetation on the cover	51
8.	MO	NITORING RADON RELEASES FROM	
	NOF	RM FACILITIES	51
	8.1.	General	51
	8.2.	Data collection at various stages	53
	8.3.	Factors in design and implementation	53
	8.4.	Quality assurance	54

9.	CAS	SE STUDY OF RADON EXHALATION	
	FRC	M A URANIUM RESIDUE REPOSITORY	54
	9.1.	Introduction	54
	9.2.	Measuring radon flux directly at the residue surface	55
	9.3.	Estimating flux by measurement of L_{μ} and E_{μ}	56
		9.3.1. In situ radon diffusion length (L_r)	57
		9.3.2. In situ radon emanation coefficient (<i>E</i>)	59
		9.3.3. Estimation	59
	9.4.	Estimating flux by calculating L_r and	
		<i>E</i> via empirical formulas	59
	9.5.	Illustrations of estimation and measurement results	60
		9.5.1. Determination of exhalation flux	60
		9.5.2. Determination of radon diffusion length	60
		9.5.3. Relating radium content to radon flux	62
APP	END	IX I: MATHEMATICAL DEVELOPMENT OF	
		RADON DIFFUSION EQUATIONS	65
APP	END	IX II: REFINED METHODS OF	
		RADON FLUX CALCULATION	71
REF	EREN	VCES	75
NOT	TATIC	DN	83
CON	ITRIE	BUTORS TO DRAFTING AND REVIEW	85

1. INTRODUCTION

1.1. BACKGROUND

Residues from the mining and milling of uranium ores as well as from a number of other resource extraction industries (for example, phosphogypsum from the phosphate industry and residues from mineral sands processing and from the oil, gas and coal industries) contain uranium or thorium series radionuclides or both [1] and are therefore naturally occurring radioactive materials (NORM). One important pathway for radiological impact arising from NORM residues is the release of radon isotopes into the atmosphere. As a result, the prediction of such releases, including the effect of various covers, is an important element in rehabilitation planning. In addition, measurement of releases from existing residue repositories is often necessary to provide an input parameter to models of radon dispersion in the atmosphere for final assessment of radiological impact and for the investigation of any required ameliorative actions.

In response to the need of its Member States, in 1992 the IAEA published Technical Reports Series No. 333 (TRS 333), Measurement and Calculation of Radon Releases from Uranium Mill Tailings [2]. TRS 333 reviewed the major aspects of radon release, control and monitoring as they relate to the management of uranium mill tailings.

Since the publication of TRS 333, a number of developments have taken place in techniques for the measurement and modelling of radon exhalation, and practical experience has been gained with various types of residue repository. Additionally, the IAEA has organized a series of projects from 1992 to the present aimed at improving environmental assessment and remediation. Through these projects, the environmental behaviour of radionuclides has been considered in many other IAEA publications, mainly in the context of contaminated site characterization and environmental remediation. These publications include: the characterization of contaminated sites [3], technical and non-technical factors relevant for the selection of the preferred remediation strategy and technology [4–6], an overview of applicable technologies for environmental remediation [7–9], options for the cleanup of contaminated groundwater [10] and planning and management issues [11]. Other IAEA publications on related aspects include reports on the remediation of uranium mill tailings [12] and of dispersed contamination [13, 14], on the decontamination of buildings and roads, on the characterization of decommissioned sites and on radiation protection and the management of radioactive waste in the oil and gas industry [15]. In view of these new publications as well as of new findings regarding radon, it was deemed necessary to produce a revision of TRS 333.

This report covers radon releases from repositories of NORM residues. For simplicity, in this report the term 'residue' includes tailings from processing, residues from heap leaching, waste rock, sludges, filter cakes and scales. Specific examples include uranium mill tailings, phosphogypsum, coal fly ash, and thorium refinery residues.

Such residues may be present in the environment in repositories in various physical forms, such as tailings ponds, dumps or stacks. Rehabilitated waste is typically covered with natural and engineered materials such as layers of clay, soil, water, rock, geosynthetics, vegetation, a combination of any of these or it may be exposed to the atmosphere.

The factors affecting radon releases from residue repositories will vary considerably with the residue type, the repository characteristics and the radon isotope of interest. This report describes approaches to the measurement, calculation and monitoring of radon releases that can be applied to many situations, together with specific examples of important representative scenarios.

There are three radon (Rn) isotopes naturally present in the environment: ²²²Rn, a member of the uranium decay series, ²²⁰Rn, a member of the thorium decay series, and ²¹⁹Rn, a member of the actinium decay series. ²²⁰Rn and ²¹⁹Rn are often referred to as thoron and actinon, respectively. In the case of actinon, its short half-life (3.98 s) and its very low activity concentration relative to ²²²Rn in the environment mean that for all practical situations its radiological impact is negligible compared with that of ²²²Rn. The impact of ²²⁰Rn is minimal, as its half-life is 55.6 s and it has a higher activity concentration than ²¹⁹Rn but still much lower relative to ²²²Rn. Consequently, the primary focus of this report is ²²²Rn, and to some extent ²²⁰Rn, wherever applicable.

1.2. OBJECTIVE AND SCOPE

This report covers the prediction, measurement and monitoring of radon releases from NORM residues, including uranium mining and milling residues. The intention is to support environmental investigations and assessments, as well as remediation in areas contaminated by NORM, by presenting state of the art concepts, models and parameters.

The primary focus of this report is the release of ²²²Rn, although ²²⁰Rn and ²¹⁹Rn are also discussed, and the material herein discussed can also be applied to those isotopes, with modifications. The report covers methods of prediction, measurement and monitoring of radon emanation from grains into interstitial spaces, transport within the residue and soil profile and exhalation into the atmosphere from the surface. The related topics of emanation and exhalation from non-residue materials, the behaviour of radon in the atmosphere following

exhalation, implications for dose assessment and applications of radon as a tracer of environmental processes are all outside the scope of this report.

This report is intended to be used in conjunction with other IAEA publications related to the assessment of the radiological impact of radioactive waste, such as Safety Reports Series No. 19, Generic Models for Use in Assessing the Impact of Discharges of Radioactive Substances to the Environment [16], IAEA Safety Standards Series No. WS-G-2.3, Regulatory Control of Radioactive Discharges to the Environment [17], and other related publications.

1.3. STRUCTURE

This report is comprised of nine sections. Section 2 concerns factors controlling radon releases from residue materials. Sections 3 and 4 describe the measurement and estimation of variables affecting the radon exhalation rate. Section 5 considers measurement methods for radon concentrations in soil gas and for radon exhalation flux density from a surface in detail, including instrumentation and its calibration. Section 6 provides methods for predicting radon exhalation flux density. Section 7 addresses waste repository cover characteristics and meteorological conditions. Section 8 briefly discusses aspects of radon monitoring programmes including data collection, factors in design and implementation and quality assurance. Section 9 outlines a case study in which multiparametric measurements at a uranium waste repository are compared with model predictions.

The report also includes two appendices presenting the main mathematical development of the radon diffusion and radon flux equations.

2. FACTORS CONTROLLING RADON RELEASES

2.1. GENERAL

The three naturally occurring radon isotopes ²²²Rn, ²²⁰Rn and ²¹⁹Rn are formed on the alpha decay of their radium parents ²²⁶Ra, ²²⁴Ra and ²²³Ra, respectively (Figs 1–3, and Table 1).



FIG. 1. Uranium decay series.



FIG. 2. Thorium decay series.



FIG. 3. Actinium decay series.

TABLE 1. RELEVANT PHYSICOCHEMICAL PROPERTIES FOR $^{222}\text{Rn},\,^{220}\text{Rn}$ AND ^{219}Rn

Parameter	Symbol	²²² Rn	²²⁰ Rn	²¹⁹ Rn
Half-life	T _{1/2}	3.8232 (8) d	55.8 (3) s	3.98 (3) s
Decay constant	λ	$2.0984 imes 10^{-6}/s$	$1.242\times 10^{-2}/s$	$1.74 imes 10^{-1}/s$
Average recoil energy on formation	$E_{\rm r}$	86 keV	103 keV	104 keV
Diffusion coefficient in air	D_{MA}	$1 \times 10^{-5} \text{ m}^2/\text{s}$		
Diffusion coefficient in water	$D_{_{ m MW}}$	$1 \times 10^{-9} \text{ m}^2/\text{s}$		

Note: Decay data are taken from Ref. [18].

Radon atoms located within solid grains are unlikely to become available for release to the atmosphere, owing to their very low diffusion coefficients in solids. However, if they are located in the interstitial space between grains, they may diffuse to the surface. Therefore, releases of radon from a residue repository to the atmosphere take place by the following series of processes [19]:

- (a) Emanation radon atoms formed from the decay of radium escape from the grains (mainly because of recoil) into the interstitial space between the grains.
- (b) Transport diffusion and advective flow cause the movement of the emanated radon atoms through the residue or soil profile to the ground surface.
- (c) Exhalation radon atoms that have been transported to the ground surface and then exhaled to the atmosphere.

These processes are illustrated in Fig. 4.



FIG. 4. Processes leading to radon release to the atmosphere.

The amount of radon available for transport to the surface is given by the radon production rate per unit volume of the residue material, P (Bq·m⁻³·s⁻¹):

$$P = \lambda E R \rho_b \tag{1}$$

where

 λ is the decay constant for radon (s⁻¹);

E is the emanation coefficient (dimensionless);

R is the radium activity concentration in the residue (Bq/kg);

and ρ_{h} is the bulk density (kg/m³).

This is the source term of radon for a particular residue material. It follows that the source term increases with the emanation coefficient, the radium activity concentration and the bulk density.

Radon in the interstitial space may be transported to the surface by diffusion and advective flow. Advective flow may be significant in some cases owing to surface cracks or holes, gas production, transport in water or the presence of large voids. However, in most cases, the dominant transport mechanism is diffusion, and therefore the transport of radon is commonly referred to as diffusion.

Following emanation and diffusion to the surface, radon may be released from the surface into the atmosphere. This release is referred to as radon exhalation, and is characterized by the exhalation flux density $(Bq \cdot m^{-2} \cdot s^{-1})$.

2.2. EMANATION COEFFICIENT

The emanation coefficient is defined as the fraction of radon atoms generated that escape the solid phase in which they are formed and become free to migrate through the bulk medium. This term has also been referred to as the emanation fraction or the emanating power.

It is generally accepted that radon escapes from a grain as a result of its recoil when its parent, radium, decays [20]. As the recoil range in solids is small, typically less than 0.05 μ m, most of the recoiling atoms remain within the mineral grain lattice. Further movement of the atoms from the lattice does not significantly contribute to radon release because of the very small diffusion coefficient through minerals, 10^{-25} – 10^{-27} m²/s [21, 22]. If the recoil terminates outside the grain or in an open pore, radon is able to migrate (Table 2 [23] and Fig. 5 [24]).

Madium	Density (z/am ³)	Projected	range (nm)
Medium	Density (g/cm ³)	²²² Rn	²²⁰ Rn
Air	1.58×10^{-3}	53×10^3	60×10^{3}
Water	1.00	77	87
Quartz (SiO ₂)	2.65	34	38

TABLE 2. ALPHA RECOIL RANGES OF ²²²Rn AND ²²⁰Rn IN VARIOUS **ENVIRONMENTS**

Note: Alpha recoil ranges were calculated using SRIM-2006 [22], assuming 86 and 103 keV of recoil energies for ²²²Rn and ²²⁰Rn, respectively.



FIG. 5. Schematic diagram of radon emanation [23].

2.2.1. Factors affecting radon emanation

Variations encountered in the emanation coefficients and the complex effects of host rock and uranium mineralogy, uranium ore grade, particle size and moisture content have been described qualitatively and by some models [24–30]. In practice, the emanation coefficient has to be measured for each material being studied. The effects of radium distribution and particle size, and of moisture and mineralogy are briefly described in Sections 2.2.1.1–2.2.1.3.

2.2.1.1. Radium distribution, particle size and shape

The particle size and shape determine in part how much uranium and radium is close enough to the surface of the grain to allow the radon to escape into the interstitial pores. If radium were uniformly distributed throughout a grain, the emanation coefficient would approach an inverse proportion to the particle diameter if this were greater than approximately 0.1 μ m. On the other hand, if radium were mainly distributed on the grain surface, the emanation coefficient would be constant regardless of the particle diameter (Fig. 6).



FIG. 6. Calculations of radon emanation coefficients in the single grain model. Radium is assumed to be distributed (a) uniformly in a spherical grain, and (b) on the surface of a spherical grain. This model does not take into account the embedding of radon into a second grain. The range of radon with a recoil energy of 86 keV in a common mineral (quartz, SiO₂) was set to 34 nm, based on the SRIM-2006 calculation [22].

Uniform distribution is often a characteristic of primary minerals [31–33], including some process residues such as fly ash [34], while surface distribution is a characteristic of secondary minerals such as sedimentary deposits and residues such as sand filters [29, 35].

There are more complex theories to explain radon emanation in situations that lie between the extremes presented above (uniform distribution vs. surface only). These theories consider factors such as fractal lattice structure [21] and adjacent particle effects [29, 30].

2.2.1.2. Moisture content

Moisture content is well known to have a strong effect on the emanation coefficient [32, 35–39]. This is because the typical ranges for recoil in water are much less than in air (see Table 2) and therefore water is more effective at stopping radon atoms within the pore space.

Figure 7 shows this effect for one type of uranium mill tailings [37]. In dry residues, the emanation coefficient is relatively low because most radon atoms escaping from particles bury themselves in adjacent particles. As the moisture content increases, the pores contain more water and a recoiling atom is more likely to terminate its recoil in the pore. At higher moisture contents (above 5% by volume), few of the atoms can penetrate into an adjacent particle and the emanation coefficient remains nearly constant with increasing moisture up to saturation. In general, emanation coefficients in saturated residues are 2–6 times higher than in dry residues. The higher value corresponds to residues with smaller particle sizes.

2.2.1.3. Mineralogy

The emanation coefficient of radon depends on a number of mineralogical aspects such as lattice structure, porosity, grain shape, and elemental composition. NORM residues are generally the product of a processing and milling process, which can have a significant effect on the mineralogy of the residue. Knowledge of the mineralogical characteristics of the residue can be important in understanding its emanation behaviour.

The emanation coefficient of soil grains is variable because soil is generally composed of many kinds of mineral grains of different rock origins. Weathering may also influence the characteristics of soils by increasing the specific surface area (area to volume ratio) of the matrix.



FIG. 7. The effect of moisture content on the ²²²Rn emanation coefficient for a sample of uranium mill tailings [37].

In the specific case of high grade uranium ore, uranium minerals are often present as thick bands or as large grains which are normally associated with low emanation coefficients. On the other hand, these high grade ores have been exposed to much higher radiation doses and this can lead to high emanation coefficients as a result of radiation damage to the crystal lattice [40]. Therefore, uranium ore grade can affect the emanation coefficient of crushed materials and their residues because of these two different and competing effects.

Residues may undergo chemical and physical changes in the repository. Over time, this can result in changes in composition and grain size [41, 42]. As a result, the method of attachment of radium within the matrix can change, which may affect the radon emanation [43, 44].

2.2.2. Radon emanation coefficients for natural materials

A large number of studies have reported radon emanation rates for natural materials including uranium mine and mill tailings, rocks, soils and minerals. As is shown in Fig. 8, radon emanation coefficients vary considerably between and within material groups. In particular, the reported coefficients for fly ash and minerals were low, generally less than 0.1, whereas they were higher for rocks, soils and mill tailings, generally lying in the range of 0.1–0.3.



FIG. 8. Radon-222 emanation coefficients measured for various materials. The area of each plot depends on the number of samples [23].

Higher coefficients reported for soil and tailings suggest that radium in these two materials tends to be enriched near the grain surface. Cracks and fissures on the surface of the grain, from previous radioactive decays and chemical or weathering effects, can also increase the surface area of the grain, thus increasing the emanation coefficient.

Emanation coefficient data for ²²⁰Rn is much sparser than for ²²²Rn. Table 3 [45–49] shows a summary of data available in the literature. In general, it can be seen that ²²⁰Rn emanation coefficients are lower than, but of a similar order of magnitude to, reported ²²²Rn emanation coefficients.

2.3. DIFFUSION COEFFICIENT

The molecular diffusion coefficient of radon is defined by Fick's first law, which states that radon flux density is linearly proportional to its concentration gradient:

Material	Sample number	Mean Rn-220 emanation coefficient	Reference
Mineral	2	0.0005 ± 0.0001	[45]
	11	0.011 ± 0.016	[46]
Average	13	0.01 ± 0.02	
Rock	5	0.157 ± 0.086	[45]
	32	0.105 ± 0.145	[47]
Average	37	0.11 ± 0.14	
Soil	12	0.11 ± 0.02	[48]
	16	0.16 ± 0.06	[49]
Average	28	0.14 ± 0.05	

TABLE 3. RADON-220 EMANATION COEFFICIENT FOR VARIOUS MATERIALS

Note: Moisture content may differ among the samples although test temperature was around room temperature.

$$f = -D_M \nabla C \qquad (2)$$

where

f is the radon flux density $(Bq \cdot m^{-2} \cdot s^{-1})$; *D_M* is the molecular diffusion coefficient (m^2/s) ;

and ∇C is the gradient of radon activity concentration (Bq/m⁴).

The negative sign arises from the fact that radon diffuses from high to low concentrations. The molecular diffusion coefficients for radon in air (D_{MA}) and in water (D_{MW}) are approximately 1×10^{-5} m²/s and 1×10^{-9} m²/s, respectively (see Table 1), and can be substituted for D_M in Eq. (2).

In a porous medium such as soil, radon moves by diffusion in the pore space between the soil particles. The rate of radon movement or flux through a soil may be slower than by diffusion in a homogeneous medium such as pure air for two main reasons:

- (a) Smaller fluid volume limiting flow (porosity, *n*);
- (b) Tortuous flow path around particles (tortuosity, τ).

Taking these factors into account, the bulk flux density of radon through a porous medium is given by:

$$f = -n_s \tau D_M \nabla C \tag{3}$$

where

f is the radon flux density $(Bq \cdot m^{-2} \cdot s^{-1})$;

- $n_{\rm s}$ is the porosity of the soil;
- τ is the tortuosity factor, which equals unity for a pure solution and is typically less than unity in soils (e.g. a typical value for τ is 0.66, as obtained for closely packed uniform spheres);
- $D_{\rm M}$ is the molecular diffusion coefficient (m²/s);

and ∇C is the gradient of radon activity concentration in the pore space (interstitial) (Bq/m⁴).

The terms τ and D_M are often lumped together to define the diffusion coefficient of the pore fluid such that:

$$D = \tau D_M \tag{4}$$

Substituting this in Eq. (3) yields:

$$f = -nD\nabla C \tag{5}$$

The quantity of nD may be represented by the notation D_e and referred to as the effective bulk diffusion coefficient of the soil. In this report, the radon diffusion coefficient D is used rather than D_e , because it is directly related to the diffusion length of radon in the matrix and is therefore the more fundamental parameter. In the simple case of a homogeneous bare residue surface, the flux is solely dependent on D and does not depend on n.

Radon diffusion coefficients in soil may be position dependent and are generally dependent on the soil type, its pore size distribution, its water content, and the degree and method of its compaction [50–52]. Because of the large difference between D_M in air and in water, moisture effects generally dominate over other physical factors. Consequently, a number of researchers have studied the dependency of radon diffusion coefficients on moisture content for different soil types [37, 53]. Such studies commonly demonstrate that the diffusion of radon is seriously hampered when soil moisture exceeds a certain threshold, which depends on the geometry of the soil pore space. Figure 9 presents measured

values for the radon diffusion coefficient, *D*, as a function of moisture content. The results show that:

- (a) For moisture saturations of less than 0.25, *D* is relatively constant, varying between approximately 9×10^{-7} and 7×10^{-6} m²/s.
- (b) As moisture saturation increases, the spread of D increases by approximately 2 orders of magnitude at intermediate saturation and by approximately 4 orders of magnitude at total saturation.
- (c) *D* decreases to values typical of saturated materials at saturations between 0.8 and 1.0.



FIG. 9. Comparison of measured radon diffusion coefficients with a simple correlation function.

The effect of moisture saturation on the diffusion coefficient can be explained qualitatively. For dry soils, the diffusion coefficient can be calculated as $D = \tau D_{MA}$. When moisture content is low, radon diffusion is not hindered by the presence of small amounts of water that form thin, discontinuous films around the soil particles, and therefore this equation still applies. At higher moisture contents, free water appears in the interstitial space between particles and radon particles must either take a longer route through the air filled pore space or else diffuse partially through water. In this region, the diffusion coefficient is not uniquely determined by the percentage moisture by weight, but depends on how the water and air phases are spatially distributed. As the porous medium approaches full saturation, the diffusion coefficient approaches the value given by $D = \tau D_{MW}$.

In the intermediate region, the diffusion coefficient is best determined experimentally, although both empirical relationships and mathematical models such as a pore distribution model can be used.

2.4. RADON EXHALATION

The exhalation flux density, denoted by f, describes the release of radon over a specific area (Bq·m⁻²·s⁻¹). The total release rate of radon, or the total radon exhalation flux, F (Bq/s), for a NORM residue facility is therefore obtained by multiplying the area averaged exhalation flux density by the total surface area of the repository. The main variables directly affecting the exhalation flux densities are:

- *R* the radium activity concentration in the pile (Bq/kg);
- ρ_{b} the bulk density of the matrix (kg/m³);
- *E* the emanation coefficient (non-dimensional);
- *D* the diffusion coefficient of radon (m^2/s) ;

and Z, the thickness of the covering material (m).

Values of these parameters are generally obtained by direct measurements, some of which are described in Section 3. If data for any of the parameters listed above are not available, then the radon exhalation flux density may be estimated by choosing default or approximate values for these parameters.

3. MEASUREMENT OF VARIABLES AFFECTING RADON EXHALATION FLUX

3.1. MEASUREMENT OF THE EMANATION COEFFICIENT

The emanation coefficient can generally be evaluated by two methods: the combination of measurements of radium and radon, and gamma spectrometry under different conditions.

In the former, for example, sample material in a cylindrical container is held in a closed vessel for more than 4 weeks to ensure the establishment of radioactive equilibrium between ²²²Rn and ²²⁶Ra (Fig. 10). After that, the total activity of radon released into the air from the sample material is evaluated by a measurement of radon concentration and the effective volume of the measurement system. Filter, Dryer, Pump, etc



FIG. 10. Schematic diagram of an example process of emanation coefficient measurement.

The effective volume is determined by considering the volume of the closed vessel, of all material inside the closed vessel, of the loop system for radon measurement and of the radon monitor. The total activity of radium in the sample material can be determined by various methods such as alpha spectrometry, gamma spectrometry, liquid scintillation spectrometry and mass spectrometry [54]. The emanation coefficient is calculated by:

$$E = \frac{VC}{MR} \tag{6}$$

where

- *E* is the emanation coefficient;
- V is the effective volume of the sampling device (m³);
- *C* is the radon concentration (Bq/m^3);
- M is the total mass of the sample (kg);

and *R* is the radium activity concentration (Bq/kg).

Another approach used for radon and radium measurements is the flow through method (Fig. 11).



FIG. 11. Schematic diagram of an example process of emanation coefficient measurement by flow through.

For evaluation, the effective volume is determined by considering the volume of a vessel and all material in the closed vessel. In addition, several hours only may be required for radon measurement because the steady state is immediately established by ventilation. Using radon free air or air with a negligible amount of radon as a carrier gas, the emanation coefficient is calculated by:

$$E = \frac{(\nu + \lambda V)C}{\lambda MR} \tag{7}$$

where

- *E* is the emanation coefficient;
- v is the flow rate (m³/s);
- λ is the decay constant of ²²²Rn (s⁻¹);
- V is the effective volume (m³);

- *C* is the radon concentration (Bq/m^3);
- M is the total mass of the sample (kg);

and R is the radium activity concentration (Bq/kg).

In the gamma spectrometry method, the sample is sealed in a container and gamma rays emitted from ²¹⁴Pb or ²¹⁴Bi are measured several times before and after the establishment of radioactive equilibrium between ²²⁶Ra and its progeny (Fig. 12).



FIG. 12. Schematic diagram of an example process of emanation coefficient measurement.

It is assumed that radon activities are always equal to those of its progeny. Gamma spectrometry is a useful non-destructive method and indirect measurement using progeny nuclides is commonly applied for radium isotopes. In addition, calibration may not be required in this method because only relative counts are needed. The emanation coefficient is calculated by:

$$E = \frac{N_{eq} - N_0}{N_{eq}} \tag{8}$$

where

E is the emanation coefficient;

 $N_{_{eq}}\,$ is the specific counts selected from peaks of $^{214}\mathrm{Pb}$ and $^{214}\mathrm{Bi}$ in equilibrium condition;

and N_{0} is the number of counts corresponding to N_{eq} in initial condition.

Analysis of multiple peaks such as 295 keV (²¹⁴Pb), 352 keV (²¹⁴Pb), 609 keV (²¹⁴Bi) and 1120 keV (²¹⁴Bi) improves precision.

In preparation for measurement, it is necessary to remove radon from the sample pores. In the case of a sample kept in a closed bottle for a long period prior to analysis, the radon progeny attached to the sample grains can also cause imprecision in the first measurement. Thus, it is common to leave such a sample (with a thickness of a few centimetres at the most) open for several hours, or to avoid the use of data measured within several hours after placing and sealing a sample into a container [55, 56].

If the ²¹⁴Pb or ²¹⁴Bi counts seem to reach equilibrium within a month, the measurement result should be corrected by the estimation of leakage impact.

3.2. MEASUREMENT OF THE DIFFUSION COEFFICIENT

Several methods are available to measure the radon diffusion coefficient in porous materials such as soil, building materials, NORM residue, etc. [19, 57–62]. Some of the methods are sample based measurements while others are of the in situ type. Since radon emissions are directly governed by in situ diffusion coefficients, the measurement of these coefficients are described below. The most popular method in the in situ category is the soil radon depth profiling method [19, 57, 58]. In this method, radon concentrations at different depths of the residue from the surface are measured by inserting a soil probe equipped with a radon monitor to generate depth profile data of radon concentration in the matrix. Concentration data collected at various depths are least square fitted to the following equation derived from diffusion theory [19]:

$$C(z) = C_{\infty} \left(1 - e^{-\frac{z}{L_r}} \right)$$
(9)

where

C(z) is the radon concentration in pore-air at depth z (Bq/m³);

 C_{∞} is the radon concentration at large depth of the residue repository (>2 m) (Bq/m³);

z is the depth (m);

and L_r is the radon diffusion length (m).

The parameters C_{∞} and L_r are obtained as fitting parameters. From the value of L_r obtained from fitting, the diffusion coefficient D_r is calculated as $D_r = \lambda L_r^2$.

Direct measurement of the effective diffusion coefficient (the product of the porosity and the diffusion coefficients) may be important for predicting radon flux, especially if information is available on the radon concentration at a great depth within the NORM residue. It can be directly estimated using the in situ accumulator technique (described in Section 5.3.1) in the context of radon flux measurements. Following the deployment of the accumulator on the residue surface, the radon concentration in the headspace increases approximately as a saturating exponential with time [63]. The time required to attain saturation in the chamber is governed by a time constant (τ_e) of the accumulator–residue system. In contrast to one dimensional diffusion models that do not predict saturation [64, 65], application of a two dimensional diffusion theory predicts saturation and indicates that the time constant is a function of the effective diffusion coefficient of radon in the matrix and the dimension of the accumulator [66, 67]. The relation is given by the following equation:

$$\tau_e = \frac{1}{\lambda + \frac{nD_r}{aH}k\chi_v} \tag{10}$$

where *a* is the chamber radius (m) and *H* is the effective height of the accumulator, i.e. the ratio of total volume (sum of volume of accumulator headspace, internal volume of 222 Rn monitor and tubing volume) to base area of the accumulator (m).

In Eq. (10)

$$\chi_{\nu} = \frac{\Gamma(\nu)\Gamma(\nu+2)}{\Gamma(\nu+0.5)T(\nu+1.5)}$$
(11)

in which $\Gamma(v)$ is the gamma function, and

$$\mathbf{v} = \frac{1}{\pi} \arccos \left[\frac{\mathbf{D}_{MA}}{\mathbf{D}_{MA} + n\mathbf{D}_{r}} \right]$$
(12)

where D_{MA} is the molecular diffusion coefficient of radon in air; and nD_r is the effective diffusion coefficient (D_r) of radon in the residue.

In view of the somewhat complicated nature of the mathematical relationship above, a look-up table is provided (Table 4) to find the effective diffusion coefficient (D_e) from the observed value of the time constant (τ_e) of accumulator set-ups of various volume to radius $(V/\pi a)$ ratios. To illustrate the use of the look-up table, consider a chamber of $V/\pi a \sim 150$ cm² for which the time constant (τ_e) is observed to be ~140 min. The effective diffusion coefficient is found by locating the nearest value of the observed time constant, 140 min, in the column representing the $V/\pi a \sim 150$ cm² from the look-up table (i.e. 141 min in the column $V/\pi a = 150$ cm²), which corresponds to an effective diffusion coefficient of 0.002 cm²/s. For intermediate values of the parameters, D_e can be obtained by linear interpolation.

TABLE 4. LOOK-UP TABLI IN THE RESIDUE MATRIX F SET-UPS WITH VARIOUS V/	E FOR FINI ROM THE (ALUES OF 1	DING THE DBSERVEI //πa (cm ²)	D TIME CO	VE DIFFU	SION COE t _e) OF RAL	EFFICIENT DON BUILI	$(D_e = nD_p)$ DUP IN ACC	OF RA CUMUL	ATOR
		i		- N/πα	(cm^2)				
	50	٢/	100	125	150	200	250	300	450
$D_{\rm e} ({ m cm^2/s})$				Time consta	int, τ_{e} (min)				
2.0E-06	1329	1840	2278	2657	2989	3542	3984	4335	5110
6.0E-06	849	1208	1534	1829	2099	2572	2975	3312	4113
1.0E-05	677	974	1248	1502	1736	2159	2527	2842	3618
2.0E-05	495	720	932	1132	1321	1670	1984	2259	2969
4.0E-05	356	522	681	834	980	1256	1511	1740	2354
6.0E-05	292	431	564	693	818	1055	1277	1478	2030
8.0E-05	254	375	493	607	717	929	1129	1311	1819
1.0E-04	228	337	443	547	647	840	1024	1192	1666
2.0E-04	162	240	317	393	467	611	749	878	1250
4.0E-04	112	166	220	273	326	429	529	623	901
6.0E-04	92	137	181	225	269	354	438	518	753
8.0E-04	79	118	157	196	234	309	382	452	660
1.0E-03	71	106	141	175	209	277	343	406	595

TABLE 4. LOOK-UP TAE IN THE RESIDUE MATRIX SET-UPS WITH VARIOUS	SLE FOR FI X FROM TH VALUES O	INDING T IE OBSERV F V/πa (cm	HE EFFEC VED TIME (1 ²) (cont.)	TIVE DIFI CONSTAN	EUSION CC T (τ _ε) OF R _l	DEFFICIEN ADON BUI	$VT (D_e = n$ LDUP IN A	<i>D</i> ,) OF R CCUMUI	ADON
				И	<i>πa</i> (cm ²)				
	50	75	100	125	150	200	250	300	450
$D_{\rm e}({ m cm}^2/{ m s})$				Time co	nstant, τ_{e} (min				
2.0E-03	48	71	94	118	141	187	232	276	407
4.0E-03	34	50	67	84	100	133	165	197	292
6.0E-03	27	41	54	67	81	107	134	159	237
8.0E-03	23	35	46	58	69	92	115	137	203
1.0E-02	20	31	41	51	61	81	101	121	180
2.0E-02	14	20	27	34	41	54	68	81	121
4.0E-02	6	14	18	23	27	36	45	54	81

4. ESTIMATION OF VARIABLES AFFECTING RADON EXHALATION FLUX

4.1. RADON EMANATION COEFFICIENT (E)

As shown in Fig. 7 (Section 2.2.1.2), the degree of saturation has a strong effect on the radon emanation coefficient in a material. However, there are some cases where materials can be expected to have significantly lower emanation coefficients, as discussed in Section 2.

The following empirical equation relates the radon emanation in dry conditions to moisture saturation [68]:

$$E = E_0 [1 + 1.85 (1 - \exp(-18.8m))]$$
⁽¹³⁾

where

E is the emanation coefficient;

 E_{o} is the radon emanation coefficient of residues in dry conditions;

and *m* is the fraction of pore space filled with water.

The emanation coefficients for ²²⁰Rn are generally lower than those for ²²²Rn (see Section 2.2.2).

4.2. DIFFUSION COEFFICIENT (D_r)

The diffusion coefficient D_r is also a sensitive function of moisture. Several correlations have been developed to evaluate this quantity from the information on moisture content. These correlations are shown in Fig. 9 (Section 2.3).

The radon diffusion coefficient in residue (D_r) may be also estimated using the empirical formula provided by Rogers and Nielson (1991) [51] that is given as:

$$D_r = D_{MA} n_T \exp(-6mn_T - 6m^{14n_T})$$
(14)

where

 D_{MA} is the molecular diffusion coefficient of radon in air; n_T is the total porosity which may be calculated from $n_T = 1 - (\rho_b / \rho_o)$;

- ρ_b is the dry bulk density;
- ρ_{σ} is the grain density of the residue matrix (~2700 kg/m³);

and *m* is the fraction of pore space filled with water and is given by the relation $m = \rho_b \theta_d / 100 \rho_w nT$, where θ_d is the moisture content on a dry weight basis and ρ_w is the density of water (kg/m³).

4.3. RADIUM ACTIVITY CONCENTRATION (R)

In most instances, the radium activity concentration in the original ore matrix will be in equilibrium with the long lived parent in that matrix. Therefore, in the absence of other data, radium concentration in the residue can be estimated from the grade of the original raw material, and any dilution or enrichment brought about by the milling and extraction processes.

An example is given here for uranium mill tailings. After processing for uranium extraction in the mill, nearly all the ²²⁶Ra will be contained in the mill tailings. There will be a certain amount of dilution in the effluent slurry that will be discharged into the impoundment owing to the addition of various process chemicals. As a result, the ²²⁶Ra concentration (Bq/kg) in the pile will be less than that in the host ore. If a dilution factor *w* (kg of solid waste for kg of uranium ore processed) is assumed, then a rough estimate of the radium concentration in the residues can be calculated from the ore grade (i.e. percentage uranium). Since 1% of uranium in the ore is equal to 1.24×10^5 Bq/kg, *R* can be obtained from:

$$R = \frac{1.24 \times 10^5 \text{G}}{w}$$
(15)

where G is the average ore grade.

5. MEASUREMENT OF RADON CONCENTRATION AND EXHALATION FLUX DENSITY

5.1. GENERAL

Before a description of methodologies for the determination of radon flux is given, it is necessary to have an understanding of the available techniques for the measurement of radon gas. The theory supporting the radon flux measurement is, for the most part, common to the various field techniques. In simple terms,
a container (also called a chamber) of suitable size is placed with the open end on the surface of the residues and the concentration increase or steady state concentration of radon in the chamber is measured. For a soil gas measurement, a hollow rod is inserted into the surface. For both the radon flux and the soil gas measurements, the radon concentration increase in the chamber or rod is measured for a specific time period.

The method selected for measuring the actual radon gas concentration depends upon a number of factors, including availability of resources and support facilities (e.g. availability of equipment), the scope of the project, the physical characteristics of the site to be investigated, the likely levels of radon concentration to be measured, time requirements and the available financial base.

5.2. MEASUREMENT OF RADON CONCENTRATION

As radon is an odourless, colourless, inert gas with isotopes of relatively short half-lives, there are no chemical or mass spectrometric techniques with sufficient sensitivity to measure it. Consequently, it must be determined using techniques that depend on the radioactive properties either of radon itself or of its decay products.

Figure 1 shows, within the uranium decay series, the radioactive decay sequence from ²²²Rn into stable ²⁰⁶Pb. Although there are several techniques for the detection of ²²²Rn and its short lived progeny, the majority of them use detection of the alpha particles from ²²²Rn, ²¹⁸Po, ²¹⁴Po or a combination of any of these, or detection of gamma rays arising from the decay of ²¹⁴Pb and ²¹⁴Bi.

An important consideration is whether or not the detection method needs to be capable of distinguishing between ²²²Rn and ²²⁰Rn. Some detection techniques such as alpha or gamma spectrometry inherently provide this capability, while techniques such as nuclear track detectors do not. In the latter case, the radon isotopes can only be distinguished through the equipment design, such as the membrane for ²²⁰Rn decay used in twin cup dosimeters [69].

5.2.1. Mode of measurement

Radon measurement techniques may be classified based on their electrical power requirements. Passive detectors do not require an electrical power source and generally rely upon the diffusion of the radon through a small gap or filter into a detection chamber. An active device requires some form of electrical power source to operate pumps, detectors or both or to create an electrical potential to enhance the collection efficiency of the detector. Radon collection and measurement methods may also be classified into instantaneous, semi-integrating (also known as continuous real time or continuous on-line) and fully integrating (also known as time averaging or time integrating) modes.

Instantaneous mode. In this mode, a sample of gas is taken on a short timescale. The sample may either be stored for later radon measurement (for example, if the sample has been taken into a previously evacuated scintillation flask), or have its radon measured immediately. An example of the latter approach is available on radon measurement equipment used in a so-called 'sniffer' mode, in which radon is typically present with minimal ingrowth of its progeny. These methods allow for a large number of measurements to be taken in a relatively short period of time.

Semi-integrating mode. In this mode, the sampling and counting are done simultaneously, and the radon concentrations are evaluated at regular intervals. Generally, the sampling is performed for periods of a few minutes to a few hours, depending on the arrangement of the measurement system. Sample collection may be carried out either by diffusion mode or by pump based flow through mode. Detection can be carried out by alpha spectrometry, ionization chamber methods or by gross alpha counting techniques. An example of a semi-integrating mode is flow through scintillation cells. Such methods are used to obtain information on rapid changes of radon concentrations in a given environment. In addition, they may be preferred over instantaneous modes owing to their superior sensitivity, reduction in the magnitude of systematic errors or their ability to measure a time varying signal (e.g. owing to diurnal variations in the system under study).

Fully integrating mode. This mode of measurement provides a time integrated radon concentration for the period of sampling (typically of the order of weeks or months). Sampling techniques operating in this mode must maintain an integrated record of each alpha particle impacting on the measurement medium. Once removed from the sampling site, they must maintain the exposure information until it is analysed. The passive methods are useful for obtaining long term averages of low levels of radon concentrations. They also have the advantage of enabling large numbers of measurements over extended regions at a relatively low cost. Examples of these methods include nuclear track detectors, electrets and the collection of radon on activated charcoal prior to measurement using techniques such as gamma spectrometry or liquid scintillation. All three types of methods discussed above consist of primary elements such as nuclear track detectors, solid surface barrier detectors, scintillation cells, electrets, activated charcoal and ionization chambers. These are briefly discussed in Sections 5.2.2–5.2.7.

5.2.2. Nuclear track detector

Nuclear track detectors are passive, integrating, lightweight and relatively inexpensive radon detectors [70, 71]. Their operation is based on the use of a selected film material, such as cellulose nitrate, polycarbonate or allyl diglycol carbonate (CR39). Alpha particles impinging on the film leave microscopic damage trails or 'tracks'. Following exposure, the film is etched in an alkaline solution (typically NaOH or KOH) to increase the size of the tracks so that the track density (number of tracks per unit of area) can be easily determined by optical microscopy or by automated scanning and counting technology. Owing to their relatively low sensitivity on short integration timescales, nuclear track detectors are more suited to soil gas radon measurements than to soil flux density measurements.

5.2.3. Solid surface barrier detectors

Use of solid state surface barrier detectors for detecting radon using alpha spectrometry is an active technique [72]. A common arrangement is to draw air through a cylindrical system with two filters in line. The purpose of the first filter is to remove the particulate activity, in particular the radon progeny present in the air. There is then a delay between the two filters in which ingrowth of radon progeny occurs. These progeny are then collected on a second filter, which faces the surface barrier detector.

Another approach involves collecting the decay products onto a metal surface placed inside a metallic chamber and maintained at a negative potential [73]. The positively charged ²¹⁸Po activity collected on the metal surface is then counted using the surface barrier detector.

In either case, knowledge of the time that the measurement system has allowed for ingrowth of the progeny in the air enables the calculation of the radon concentration.

An advantage of alpha spectrometry is the ability to spectrally separate the signals of the ²²²Rn and ²²⁰Rn progeny as these have different alpha energies. Surface barrier detectors have very low intrinsic backgrounds, enabling a low detection limit to be attained. Other advantages include their compact size, high energy resolution and ruggedness. Disadvantages include the necessity of an electrical power supply and associated electronic equipment and the relatively high cost of alpha spectrometry detection systems.

5.2.4. Scintillation cell

Scintillation cells are enclosed cylinders internally coated with a scintillation material, generally silver activated zinc sulphide powder, with a transparent end window. Interaction of alpha particles with the scintillation material causes production of light photons, which are detected and amplified by a photomultiplier adjacent to the end window, and converted to a count rate by an electronical device.

Scintillation cells can be operated in a closed or flow through arrangement. Commonly, the air passes through a filter prior to entry to the cell, especially in flow through systems; this is to remove dust as well as radon progeny present in the inlet air. The removal of the short lived radon progeny allows a reliable calculation of their subsequent ingrowth in the measurement system.

It is possible to distinguish ²²²Rn and ²²⁰Rn using scintillation cells, either by the use of two cells with a delay line for ²²⁰Rn decay [74], by programmed counting [75] or by use of the pair counting technique to identify the successive alpha particles produced from ²²⁰Rn and ²¹⁶Po [76].

Scintillation cells are amenable to good control of measurement uncertainties, meaning that high accuracy can be achieved. Cells are also robust and relatively low cost, and they allow the rapid assessment of ²²⁰Rn concentration. A disadvantage of scintillation cells is that with use the background slowly increases owing to buildup of ²¹⁰Pb and its alpha emitting progeny ²¹⁰Po within the cell. This may be solved by periodically replacing the zinc sulphide screen. Another disadvantage is that dual measurement of ²²²Rn and ²²⁰Rn using the pair counting approach is limited to situations where the radon concentration is much lower than the ²²⁰Rn concentration.

5.2.5. Electret

The electret radon detector is an electrically charged Teflon disc serving both as a source of an electrostatic field and as a sensor. Alpha decays in the detector chamber cause ionization of the air in the chamber and a consequent decrease of charge on the electret. The electret voltage drop over the measurement period is used to quantify the radon concentration.

Electrets are passive, integrating, lightweight and relatively inexpensive radon detectors. Their disadvantages include the necessity of allowing for the discharge of voltage caused by gamma radiation [77] and the potential influence of humidity condensation in the detector chamber on electret discharge [78, 79].

5.2.6. Activated charcoal

Activated charcoal is used as a passive medium on which radon can be adsorbed. The charcoal is weighed into a canister and sealed until it is exposed to the air to be tested. The period of exposure can range from a few hours to several days, depending on the anticipated level of radon to be encountered and the required detection limit. Following exposure, the canisters are again sealed and the radon activity determined, most commonly by gamma spectrometry, although liquid scintillation can be used following radon extraction [80].

Charcoal canisters are simple, passive, robust and low cost, and allow a relatively large number of measurements to be made over a reasonable time period (i.e. of a few days). Their capacity to adsorb radon is reduced by adsorption of water, so they are less suitable for wet or humid environments.

5.2.7. Ionization chamber

Ionization chambers used for measuring radon are enclosed metal cylinders constructed with a central collecting anode. They are accurate measurement systems that operate on the principle that the alpha particle energy dissipated during the natural radioactive decay of various radon daughters contained within the cylinder is collected on electrodes and measured as an ion current by means of an electrometer. These units can be operated in closed or flow through mode and, depending on the background and the chamber volume, they can achieve detection limits as low as 4 Bq/m³.

5.3. MEASUREMENT OF EXHALATION FLUX DENSITY

The three basic approaches for the measurement of radon exhalation flux density are accumulation, flow through systems and adsorption. Each of these is described below.

5.3.1. Accumulation

The accumulation method is commonly used for measuring the exhalation flux densities of ²²²Rn, ²²⁰Rn and also of stable trace gases such as CO_2 and CH_4 [81]. It involves the placement of a chamber known as an accumulator [63, 82, 83], closed at one end and open at the other, inverted on the surface of interest. In order to prevent superficial leakage of accumulating radon, the mouth of the chamber is sealed onto the surface by inserting a short length of its cylindrical wall into the residue matrix. The radon atoms exhaled from the residue surface underlying the chamber

enter the headspace resulting in a gradual buildup of the radon concentration. This concentration is measured at several regular intervals during the deployment period or, in some cases, just once at the end of the deployment period [84]. From these measurements, the exhalation flux density is quantified. The radon concentration is measured using on-line monitors or instantaneous methods discussed in Section 5.2, such as scintillation cells or alpha spectrometry.

Figure 13 shows a schematic diagram of the set-up for a cylindrical accumulator. The concentration of radon inside the accumulator at a time *t* after deployment depends on the volume, *V* in m³ (= $V_1 + V_2$, where V_1 is the accumulator headspace volume and V_2 is the volume of the measuring device connected on-line), the surface area of the opening of the accumulator $A(m^2)$, the radon isotope decay constant λ (s⁻¹), the radon exhalation flux density *f* (Bq·m⁻²·s⁻¹), the degree of back diffusion of radon and the degree of ventilation in flow through mode operations. Back diffusion refers to the loss of radon from the accumulator via the soil to the atmosphere at the soil–accumulator interface and is especially significant for smaller volume (≤ 1 L) accumulators [66]. Possible leaks and adsorption on the accumulator material may also affect the concentration measurement.



FIG. 13. Schematic diagram of the use of an accumulator.

5.3.1.1. Selection criteria of the accumulator

An accumulator can be a single chamber device that is directly inserted into the surface or it can be a two part device — a collar that is inserted into the soil and a cover that is then sealed onto the collar [83]. The use of radon absorbing materials should be avoided. The accumulator dimensions can vary depending upon the flux levels, portability requirements, time available for a measurement and desired resolution to map the flux across a given surface. A large base accumulator (~20–50 cm in diameter) is useful for obtaining representative data over larger areas and allows better coverage of a heterogeneous surface (rocks, burrows, vegetation, etc.). In addition, larger bases allow more radon to enter the chamber and hence are useful when the flux levels are small. However, large accumulators are not convenient from the point of view of portability, especially for large scale field studies.

Smaller accumulators (around 5–10 cm in diameter) are useful for high resolution spatial measurements to explore local variations of fluxes. However, they have larger back diffusion effects that affect the flux; these effects need to be accounted for through process based models. Similar considerations also exist regarding the height of the accumulator. For taller accumulators, radon may not be mixed uniformly at all heights and it may be necessary to employ a small fan inside the chamber for mixing. On the other hand, back diffusion effects will be stronger in chambers with smaller heights. In actual practice, the choice of the accumulator diameter and height should be based on considerations of portability, flux levels, spatial representativeness of measurements and the sensitivity of the monitoring instruments.

5.3.1.2. Model for interpreting concentration data

The mixing scenario in the air space bounded by the accumulator changes from a state of uniform mixing prior to deployment (due to ever present atmospheric turbulence) to a state of diffusive mixing (due to isolation from the open atmosphere) post-deployment. This causes an initial flux drop by a factor $k = [1 + n (D/D_a)^{0.5}]^{-1}$, (where n, D are the porosity and radon diffusion coefficient in the residue matrix and D_a is the molecular diffusion coefficient of radon in air). The value of k is typically about 0.88. Further, the back diffusion effect brings about a gradual drop in the flux as the concentration builds up inside the chamber. This is modelled by two dimensional diffusion theory [67] according to the formula:

$$C(t) = C_0 + C_m \left(1 - \exp\left[-t/\tau_e\right] \right) \tag{16}$$

where

- *C* is the radon concentration;
- *t* is the measurement time;
- *Cm* is found by $k f (A/V)\tau_{e}$, where f is the radon flux density per bulk area, A is the surface area of the opening of the accumulator, V is the effective volume of the sampling device (m³);
- C_0 is the initial radon concentration inside the accumulator (at t = 0), which is assumed to be the same as the steady atmospheric radon concentration (Bq/m³);

and τ_{e} is the time constant of radon buildup.

 τ_e has a complicated dependence upon the dimensions of the chamber and the radon diffusion coefficient *D* in the residue. It can vary from a few minutes for small chambers to several hours for large chambers. Upon fitting the data on radon concentration measured at regular intervals (at least 6 points) to Eq. (16) the parameters $C_{q_e} C_m$ and τ_e can be estimated, from which the flux, *f*, may be estimated.

For measurement durations where t is significantly lower than τ_{e} , Eq. (16) approximates to a linear formula:

$$C(t) \approx C_0 + k \frac{A}{V} ft \tag{17}$$

The flux may be estimated from the slope obtained by linear regression analysis of the concentration data with respect to time. Equation (17) is ideally suited for short duration accumulator deployment during which only a few concentration measurements have been made.

The method of data analysis for measurement of ²²⁰Rn flux is somewhat different from that for ²²²Rn. In view of its short half-life, the ²²⁰Rn concentration in the accumulator will attain a steady state within approximately 5 min. If C_1 is the steady state concentration, the flux may be estimated with Eq. (18).

$$f_{\rm Rn-220} = \frac{V\lambda_{\rm Rn-220}}{A} (C_1 - C_0)$$
(18)

5.3.2. Flow through method

In the flow through method, a chamber working on a similar principle to an accumulator is placed over the surface to be investigated. However, in this case,

the air in the chamber is continuously removed at a constant rate and the radon concentration in the exhaust stream is measured by a semi-integrating technique such as a flow through scintillation cell.

The flow rate of air through the chamber must be sufficiently high to prevent buildup of radon in the headspace of the chamber (i.e. a steady state should be achieved). However, if the airflow rate is too high, the radon concentration in the exhaust air will be low and cause increased uncertainties in the radon measurements. The air pressure in the chamber should be the same as the ambient pressure outside. Care should be taken to ensure that restricted replacement of withdrawn air or use of a pump at the inlet do not result in lower or higher pressure, respectively.

At steady state, the exhalation flux density is proportional to the concentration of radon in the exhaust stream and to the exhaust stream flow rate, and inversely proportional to the active surface area of the chamber. Other factors in the calculation of radon flux density are dependent on the details of the system and its calibration.

For non-spectroscopic radon detection such as scintillation cells, the presence of ²²⁰Rn and its decay products must be allowed for, often by use of a delay line for ²²⁰Rn decay prior to filtration and counting. Double cell systems have been developed for simultaneous measurement of both radon and ²²⁰Rn exhalation flux densities [74]. Figure 14 shows a schematic diagram of such a double cell flow through system. In this example, a filter is used to remove radon and ²²⁰Rn progeny from the airstream, and radon and ²²⁰Rn together with their subsequently ingrown progeny are counted in the first flow through scintillation cell. The delay line allows the decay of most of the ²²⁰Rn. A second filter then removes the radon and ²²⁰Rn progeny, and the second scintillation cell counts radon and its subsequently ingrown progeny together with a small component of the remaining ²²⁰Rn and ²²⁰Rn progeny.



FIG. 14. Schematic diagram of the use of a double cell flow through system.

5.3.3. Adsorption

The adsorption method for ²²²Rn exhalation flux involves the use of an adsorption medium (usually activated charcoal) placed in close proximity to the surface being investigated.

Figure 15 shows a schematic diagram of a charcoal adsorption canister. Before use, the charcoal is heated in an oven to remove any radon, moisture and other contaminants, which may have been adsorbed previously. This drying may be carried out when the charcoal is already in the canister or prior to it being weighed into canisters. Once prepared, the canister is sealed to prevent adsorption of ambient radon or moisture onto the charcoal.



FIG. 15. Schematic diagram of the use of a charcoal adsorption canister.

Following exposure, the canisters are again sealed and the activities of the radon progeny 214 Pb and 214 Bi measured, most commonly by gamma spectrometry, following a short (~3 h) ingrowth period for the progeny. Liquid scintillation counting may be used as the measurement technique if a higher counting efficiency is required [80]. The measurement should take place as soon as practicable after the progeny ingrowth period in order to minimize loss due to radon decay.

The radon exhalation flux density over the period of exposure can be estimated using the expression:

$$f = \frac{N \cdot t_c \cdot \lambda^2 \cdot \exp(\lambda t_d)}{\varepsilon \cdot A \cdot [1 - \exp(-\lambda t_e)] \cdot [1 - \exp(-\lambda t_c]}$$
(19)

where

- f is the radon flux density $(Bq \cdot m^{-2} \cdot s^{-1})$;
- N is the net count rate, after background subtraction, obtained during the counting period (counts per s, or s^{-1});
- $t_{\rm c}$ is the counting period (s); λ is the radioactivity decay of
- λ is the radioactivity decay constant for ²²²Rn (s⁻¹);
- t_d is the delay period from the end of the exposure to the beginning of the counting interval (s);
- ε is the counting efficiency of the system relative to the activity of adsorbed radon (Bq/s);
- A is the area of the canister (m^2) ;

and t_{a} is the period of exposure of the charcoal in the canister(s) [85].

When exposed to humid or wet environments, the charcoal will adsorb moisture, which can lead to reduction in the radon collection efficiency [80, 86]. To ensure that water adsorption is not problematic or to allow any required correction, water adsorption can be quantified by weighing the canisters before and after exposure.

5.3.4. Measurement of soil gas concentration

Soil gas measurement is normally carried out using a tube inserted into the surface. This probe is open at the end, or has an opening at the side, to allow the diffusion of radon into the inside volume. The radon concentration inside the probe is measured either in situ (for example, using nuclear track detectors) or by removal of a sample of the air inside the probe for measurement of radon by an instantaneous or continuous on-line method [87, 88].

Depending upon the capabilities and requirements of the measurement technique employed, the system may need to include a porous membrane to ensure removal (by decay) of 220 Rn, a dehumidifier or a filter, or both to remove radon progeny. As radon concentrations in soil gas are generally in the range of 10^3-10^5 Bq/m³, and concentrations in NORM residue may be higher than this,

the upper end of the calibration range of the measurement instrument may need consideration.

It is important that a good seal is obtained between the probe and the soil or residue being tested. This may be more difficult to achieve where stones are present. One study [88] indicated that passive methods may give lower results than active methods for largely impermeable clay soils, due to decay of radon during diffusion from the soil pores.

Soil gas radon concentrations exhibit seasonal changes [89–91]. Diurnal variations have also been observed for soil layers close to the surface [92].

5.3.5. Measurement of mass exhalation rate

The mass exhalation rate is defined as the radon activity released into the air per unit time from the mass of the matrix and is measured by enclosing the sample in a closed chamber and monitoring the buildup of radon concentration in the chamber at regular time intervals. Typically, about 350–500 g of residue sample is enclosed in a leaktight metallic chamber coupled to a continuous ²²²Rn monitor. The radon concentration C(t) at time t since the closing of the chamber builds up according to the formula [93–95]:

$$C = \frac{J_m M}{V \lambda_e} \left[1 - e^{-\lambda_e t} \right] + C_0 e^{-\lambda_e t}$$
⁽²⁰⁾

where

- J_m is the mass exhalation rate (Bq·kg⁻¹·h⁻¹);
- is the ²²²Rn concentration present in the chamber volume at t = 0 (Bq/m³);
- M is the total dry mass of the sample (kg);
- *V* is the effective volume (volume of chamber + internal volume of ²²²Rn monitor volume of sample) (m³);
- λ_e is the effective decay constant for ²²²Rn, which is the sum of the leak rate (if existing) and the radioactive decay constant of ²²²Rn (s⁻¹);

and t is the measurement time (s).

Upon least square fitting of the data to the above formula, J_m may be obtained from the fitted parameters with the knowledge of the dry mass M of the sample. The dry mass is estimated by drying the sample at about 110°C until it attains constant weight.

5.4. CALIBRATION AND UNCERTAINTIES

The assurance of accuracy expected in measurement results is realized by traceability. That is, the status of the calibration method for utilized monitors should be in line with a national or international standard. In addition, procedures for calibration and performance tests should be described clearly and objectively in the written quality assurance plan established by each organization.

Generally, the instrumentation for field measurements should be tested and calibrated using a STAR (System for Test Atmospheres with Radon) operated by reference organizations or manufacturers. Since a STAR provides a reference concentration of radon in the air, this represents a testing and calibration of the equipment's system for measuring radon concentration only. The radon concentration in the reference atmosphere is usually determined by a secondary standard, which is calibrated by the primary standard. The calibration is performed under atmospheric standard conditions, i.e. 20°C, 60% relative humidity and 101.3 kPa atmospheric pressure.

The air sample should be filtered to remove ²²²Rn and ²²⁰Rn decay products before entering the detector. This is a requirement not only for continuous methods but also for grab sampling methods, owing to the restraint of increasing background.

Recently, radon measured absolutely has become popular as a primary standard in some reference organizations [96, 97]. However, radon released from an aqueous solution of a soluble ²²⁶Ra salt is also available and may have the benefit of reproducibility.

For ²²²Rn exhalation flux density measurements, the greatest measurement uncertainty is the degree of uniformity of mixing in the collection chamber and measurement instrument. Therefore, in this case, it is extremely important to include a mixing element in the uncertainty budget.

For accurate field measurements, instrumentation should be protected against more severe environmental conditions than are found in laboratories. In addition, performance tests must be carried out in expected environmental conditions because it is difficult to remove all confounding factors. The major considerations in special requirements for radon flux density measurements are given below.

Detection limits. The measurement plan should not only consider the lower detection limit but also the upper one. The radon concentration in the accumulation chamber can be expected to sometimes be higher than several hundred thousand Bq/m³. For example, for a typical radon exhalation flux density from uranium mill tailings of 10 Bq·m⁻²·s⁻¹, the radon concentration in an accumulation chamber of 10 cm in height reaches approximately 4×10^5 Bq/m³ after 1 h.

Background. Even in the continuous measurement of indoor radon, ²¹⁰Pb, a long lived decay product of radon, accumulates in the detector and may cause uncertainty in the radon measurement. For radon exhalation flux density measurement, the ²¹⁰Pb in the detector should be checked periodically and cautiously owing to higher radon concentrations in the accumulation chamber than in the indoor environment.

Detector volume. The volume of the detector, including sampling devices such as pumps and tubes, may also cause uncertainty in estimation of the activity concentration of radon, which is accumulated within a given time in a container with a known volume.

Sensitivity to ²²⁰Rn. Generally, ²²²Rn and ²²⁰Rn are separated by using their very different half-lives. Therefore, especially in continuous measurements, special attention should be paid to the half-life of ²²⁰Rn. Alternatively, ²²²Rn and ²²⁰Rn can be separated using the alpha particle energies emitted from them or their decay products or both.

Humidity. The humidity inside the accumulation chamber may affect the efficiency of some detectors or detection systems, such as ionization chambers, activated charcoal, electret, etc. Air drying systems may be available for some instruments.

Gamma dose. The ambient gamma dose may affect the efficiency of some detectors or detection systems, such as ionization chambers and electrets.

A quality assurance plan may be established, applied and maintained to objectively show the consistency of measurement results or the compliance with relevant laws, regulations and other requirements or both. Such a plan should make clear its scope, satisfy laws and regulations, and be accepted by all interested parties. The plan could also include a standardized system in writing, such as a procedure for measurement, handling and for the maintenance of instruments used, recording and so on. These should be included in the monitoring plan described in Section 8 in detail.

6. SIMPLIFIED EXPRESSIONS FOR ESTIMATING RADON EXHALATION FLUX DENSITY

6.1. GENERAL

The control of radon releases from radium bearing NORM residue repositories begins with an evaluation of the radon source term and the effectiveness of any cover introduced to attenuate radon migration to the surface. The main factors affecting the generation, transport and control of radon in residue matrices have been discussed in general terms in Section 2. In the present section, mathematical expressions are provided to enable simple calculations for estimating the radon exhalation flux density.

The simplest approach for estimating exhalation flux density is based on the application of diffusion theory to radon transport in the repository matrix, under the following idealized assumptions:

- (a) The residue properties are homogeneous.
- (b) Radium is uniformly distributed in the residue matrix.
- (c) Transport owing to pressure induced advection is negligible.

Although certain variations in the residue properties and radium content are inevitable, representative values of these parameters are used to evaluate local fluxes. If these variations are essentially limited only with respect to depth (homogeneous laterally), the multilayer approach can be used to evaluate the fluxes. Further, assumption (c) is generally justified under the premise that it is difficult to sustain appreciable pressure differentials between the soil and the open atmosphere for a significant length of time. Given these assumptions, expressions are provided for three situations, namely: radon flux from a bare residue repository, a repository covered with homogeneous material (with negligible radium content) and a repository with a multilayer cover. The development of the radon diffusion equation is advanced in Appendices I and II.

6.2. RADON EXHALATION FLUX DENSITY FROM BARE RESIDUE

The mean radon diffusion length, which may simply be referred to as the diffusion length, L_r , is a measure of the mean distance travelled by radon atoms in the residue matrix before their radioactive decay. It is defined as:

$$L_r = \left(\frac{D_r}{\lambda}\right)^{1/2} \tag{21}$$

where D_r is the radon diffusion coefficient of the residue (m²/s); and λ is the decay constant of ²²²Rn (s⁻¹).

 L_r is a measure of penetration by a plume into a zone with diffusion constant D_r owing to a constant concentration source at the origin, with a larger L_r corresponding to greater penetration. If the concentration source is unity, L_r is the square root of the centroid of that plume with respect to the origin.

For bare residues, there are two general cases. In the first case, the residue thickness is much greater than the radon diffusion length and the residue may be regarded as being of infinite thickness for the purpose of the estimation of radon flux density. In the second case, the thickness of the residue repository is comparable to or less than the diffusion length.

6.2.1. Residue thickness much greater than radon diffusion length

In the case of the residue thickness being much greater than the radon diffusion length, diffusion theory predicts that the radon flux density is directly proportional to the radium activity concentration, the emanation coefficient and the bulk density, and has a square root dependence on the radon diffusion coefficient:

$$f_r^{\infty} = R\rho_b E \left(\lambda D_r\right)^{1/2} \tag{22}$$

where

- f_r^{∞} is the radon flux density;
- *R* is the radium activity concentration;
- ρ_{h} is the bulk density;
- *E* is the emanation coefficient;
- λ is the decay constant;
- D_r is the radon diffusion coefficient;

and the subscript 'r' refers to the residue material and the superscript ' ∞ ' refers to a residue whose thickness is large (the expression is strictly true only in the limit as the thickness approaches infinity).

Equation (22) is also applicable to ²²⁰Rn, in which case the flux is essentially contributed by the ²²⁴Ra content present in the top layer of a few cm thickness, which is about the order of a diffusion length for ²²⁰Rn. For a typical top residue with a D_r value of 2 × 10⁻⁶ m²/s, the diffusion length for ²²⁰Rn is approximately 0.013 m.

6.2.2. Residue thickness comparable to or less than the diffusion length

When the thickness of the residue repository is comparable to or less than the diffusion length L_r (~1 m for ²²²Rn), Eq. (22) requires modification. For a residue repository of thickness z_r , diffusion coefficient D_r (diffusion length L_r) and porosity n_r resting on a soil with porosity n_s and diffusion coefficient D_s , the solution of the multilayer radon diffusion equation (see Appendix I) yields the following expression for the flux density, f_r , of radon on top of the repository:

$$f_r = f_r^{\infty} \frac{2\sinh\left(\frac{z_r}{2L_r}\right) \left[\sinh\left(\frac{z_r}{2L_r}\right) + k\cosh\left(\frac{z_r}{2L_r}\right)\right]}{\sinh\left(\frac{z_r}{L_r}\right) + k\cosh\left(\frac{z_r}{L_r}\right)}$$
(23)

where

$$k = \frac{n_r}{n_s} \left(\frac{D_r}{D_s} \right)^{1/2}$$

Three limiting formulas can be obtained from Eq. (23).

If k is much lower than 1 ($k \ll 1$), i.e. if the matrix below the residue repository is porous, and has a diffusion coefficient and porosity that are much greater than those of the repository, the following expression is obtained:

$$f_r = f_r^{\infty} \tanh(z_r/2L_r)$$
 for a porous base $(k=0)$ (24)

On the other hand, if the base of the repository is totally impervious to radon $(k = \infty)$, the equivalent expression is (see Appendix I):

$$f_r = f_r^{\infty} \tanh(z_r/L_r)$$
 for an impervious base $(k = \infty)$ (25)

The impervious base is the most conservative estimate that yields the highest flux since penetration of radon into the soil–water matrix below the repository will always reduce the emission. The variation of the flux as a function of $z_r/2L_r$ for different k values is shown in Fig. 16.

As may be seen from Eq. (23), residue thickness beyond $2L_r$ (~2–4 m) has little influence on flux. For shallow repositories ($z_r \ll L_r$) with impervious bases, Eq. (24) suggests that all the radon escapes to the surface. When z_r is much greater than L_r ($z_r \gg L_r$), Eq. (23) is equivalent to Eq. (21).



FIG. 16. Variation of the normalized radon flux (f_r/f_r^{∞}) from a residue repository as a function of non-dimensional residue thickness $(z_r/2L_r)$ for different values of the parameter k.

6.3. RADON FLUX DENSITY FROM RESIDUE REPOSITORIES WITH A SINGLE COVER

Earthen materials with a low radium content are effective covers for attenuating radon exhalation from residue repositories. If radon emission from the cover materials is neglected, the flux density from a residue surface covered with soil of thickness z_c may be approximated by:

$$f_{cr} \approx f_r \exp\left[-z_c/L_c\right] \tag{26}$$

where $L_{\rm e}$ is the radon diffusion length in the cover material. Since $L_{\rm e}$ decreases significantly with increasing moisture content in the material, a cover material that compacts well and retains moisture is the most effective for controlling radon releases.

The following example illustrates the application of Eqs (22) and (26). For a moist uranium residue pile with $R = 4 \times 10^4$ Bq/kg, $\rho_b = 1500$ kg/m³, E = 0.2and $D_r = 1.0 \times 10^{-6}$ m²/s, which corresponds to $L_r = 0.69$ m, the radon flux density is estimated using Eq. (22) as 17 Bq·m⁻²·s⁻¹. This calculation is applicable for repositories of thickness exceeding about 2 m.

For the example above, the ratio of the surface flux to radium activity concentration (*R*) is equal to $4.3 \times 10^{-4} \text{ kg} \cdot \text{m}^{-2} \cdot \text{s}^{-1}$ or, if *R* is expressed in Bq/g, this ratio is 0.43 g \cdot \text{m}^{-2} \cdot \text{s}^{-1}. For moist uranium residues with $D_r = 2.5 \times 10^{-7} \text{ m}^2/\text{s}$, the ratio is 0.22 g $\cdot \text{m}^{-2} \cdot \text{s}^{-1}$. It has been generally found that the ratio of flux (Bq \cdot m⁻² · s⁻¹) to radium concentration (Bq/g) increases with decreasing moisture, ranging from 0.2 for wet residues to 0.6 for moist residues and 1.2 g \cdot m⁻² · s⁻¹ for dry residues [98].

If the same repository is covered by 1.5 m of soil with $D_c = 4 \times 10^{-7} \text{ m}^2/\text{s}$, the ²²²Rn flux density decreases by a factor of approximately 31 (i.e. to a value of 0.6 Bq·m⁻²·s⁻¹).

6.4. RADON FLUX DENSITY FROM RESIDUE REPOSITORIES WITH MULTIPLE COVERS

It may be advantageous or necessary to use different materials to cover a residue repository. This situation requires multilayer analysis. Multilayer analysis is also useful to address situations where the moisture varies significantly with depth for a single material cover, thereby causing drastic variation in the diffusion coefficients with depth. A three layer system is shown schematically in Fig. 17. Each layer is characterized by a diffusion coefficient (and equivalently by a



FIG. 17. Multilayer cover system on the surface of the NORM residue.

diffusion length) and thickness. The value of the diffusion coefficient depends upon the moisture content, compaction and type of material.

As a first approximation, Eq. (26) can be applied to each cover layer successively considering the radon flux density from the lower layers to be the source flux into the cover layer being analysed. This gives:

$$f_{c1} \approx f_r \, \exp\left[-z_{c1}/L_{c1}\right]$$

and

$$f_{c2} \approx f_{c1} \exp\left[-z_{c2}/L_{c2}\right] \approx f_r \exp\left[-\left\{z_{c1}/L_{c1}\right\} - \left\{z_{c2}/L_{c2}\right\}\right]$$

This may easily be generalized to an *n*-layered system as

$$f_{c1} \approx f_r \exp\left[-\sum_{j=1}^{i} \left\{-z_{cj}/L_{cj}\right\}\right]$$
 (27)

The factor multiplying f_r on the right hand side of Eq. (27) is the effective attenuation factor for a combination of multiple covers. The usefulness of these equations may be demonstrated by considering the previous single layer example to be actually stratified into two layers (i = 2) of widely differing moisture content and hence differing diffusion coefficients. If the lowest metre of the cover has a diffusion coefficient of 4×10^{-7} m²/s and the uppermost half metre has a diffusion coefficient of 2×10^{-6} m²/s, then, from Eq. (27), the exhalation flux density is calculated as $f_{c2} = 1.1$ Bq·m⁻²·s⁻¹.

7. EFFECTS OF METEOROLOGICAL CONDITIONS AND THE REPOSITORY COVER

7.1. METEOROLOGICAL EFFECTS

The weather parameters of air pressure, temperature, rainfall and wind speed affect radon exhalation flux from the surface of residue repositories. The physical and hydrological properties of the cover and residue and the length of time for which radon fluxes are reported are important in assessing the extent of meteorological effects. In particular, diurnal and seasonal variations in the weather parameters can lead to variations in flux over these time periods.

Usually, the duration of atmospheric pressure changes is much less than the half-life of ²²²Rn. The exhalation flux density from the soil surface can be expected to increase if the atmospheric pressure decreases. Since atmospheric pressure changes are cyclic, the increases and decreases in the flux tend to compensate for each other. Thus, the long term fluxes are not strongly dependent on atmospheric pressure changes.

Strong winds can cause erosion and are of particular concern for exposed residues, where contaminated particles may be carried to surrounding locations. In cold regions, strong winter winds have contributed to freeze-drying effects on covers and residues. Where residue has been isolated under a water cover, wind induced wave action may increase radon transport to the surface by mixing or reducing the cover thickness.

Longer term seasonal variations may influence the annual average radon fluxes. For example, a snow covering and frozen ground can decrease the annual surface flux. Rainfall directly affects soil moisture content. In one study in a tropical region, radon exhalation flux densities were observed to increase as the soil started turning from dry to moist, followed by a rapid decrease as the rainy season started. Average radon exhalation flux densities over the wet season were lower than in the dry season, but had greater variability [99].

Similarly, long term temperature changes may alter the flux. For example, higher temperatures can slightly increase the diffusion coefficient, which slightly increases the flux. Although the surface temperature of a large residue pile can fluctuate widely, the temperature within the pile is typically more uniform. However, depending on the physical and chemical properties of the pile, other processes may lead to high internal temperatures (e.g. bacterial reactions).

Covers that are used to reduce radon emissions are often based in part on consideration of local meteorological conditions. Covers that depend on evapotranspiration to help control infiltration rely on seasonal fluctuations in precipitation and temperature to perform effectively. Higher temperatures in a waste rock pile may lead to early degradation of geomembrane liners. Long term modelling of cover systems should consider the impact of a changing climate on the integrity of a cover system.

7.2. ASPECTS OF REPOSITORY COVER DESIGN

7.2.1. Surface cover effects

During the remediation of mining and industrial sites, covers or caps are often placed on the surface of the residue to control the transport of contaminants to the environment. One important design consideration for cover systems is their ability to adequately reduce radon exhalation. Regulations may vary, requiring for example that a cover achieve a prescribed exhalation flux density value, or that radon levels be reduced to their background levels.

The performance of a cover system component depends on two factors: (1) the performance of the component itself and (2) the way individual components interact as a system. Owing to the long half-life of the radon progenitors, covers are typically expected to perform over extended time frames. Several factors such as wind and soil erosion, changes in moisture content and intrusion by tree roots and burrowing animals may result in changes in performance over time. Given the absence of observations and performance data in the long term, models are used to predict the long term performance of residue containment systems.

Covering layers reduce radon flux density by increasing the time taken by radon atoms produced in the residue to be transferred to the surface, allowing for radon to decay before reaching the atmosphere. Radon migration through the cover material is largely governed by diffusion, although advective processes may also assist in the transport, e.g. if pressure differentials exist across the material. Thus, decreasing the radon gas diffusivity and permeability of the cover media can increase cover efficiency. The diffusivity and permeability of the soil generally depend on orientation, soil type, pore size distribution, water content and the degree and method of compaction.

The degree of diffusive transport of radon through any material can be characterized by a quantity known as the mean diffusion length, or relaxation length, which can be deduced from the ratio of the diffusion coefficient to the radioactive decay constant (see Ref. [100] for an exact characterization), with greater radon penetration at steady state. This quantity can vary greatly between materials. For fine grained moist soils, the diffusion length for ²²²Rn ranges from 0.2 to 0.5 m, whereas for coarse, dry sands, the value may be as high as 1.5 m. Conversely, in dense material (e.g. dense granite), the diffusion length is 2.2 m. The diffusion length of ²²⁰Rn is about 77 times smaller than that of ²²²Rn for any given material.

In the following subsections, considerations for radon migration related to different aspects of repository cover design will be described.

7.2.2. Surface cover materials

Cover materials that have been effective in reducing radon emissions include water, earthen materials, geosynthetics such as geomembranes and geosynthetic clay liners, and evapotranspirative barriers. Simple covers may contain one type of material; however, combinations of different materials are often required. Factors that influence the selection of cover material include climate, availability of materials, properties of the residue, regulatory requirements and other site specific needs or issues (e.g. generation of acid rock drainage, water infiltration, erosion).

7.2.2.1. Water covers

As the diffusion coefficient of radon is much lower in water than in air, water covers can significantly reduce radon emissions. Subaqueous disposal is more common at uranium mine waste facilities than at other NORM facilities. Underwater disposal is typically achieved by flooding a residue impoundment or relocating the residues or waste rock to a storage basin such as a former open pit mine. In shallow water covers, diffusion may become enhanced by natural convection, wind action and evaporation. Ref. [101] examines the issue of surface water effects in predicting radon emissions from water covered uranium residue impoundments, where residue characteristics and local topography can vary significantly. For the shallow (<1 m) impoundment water, water movement was sufficient to remove radon from the residues-water interface and transport it to the atmosphere in a short time (several hours), resulting in virtually no radon containment by the surface water itself compared with similar bare saturated residues. Nevertheless, the radon flux from residues saturated with water is still much less than that from dry residues [37]. Environments with snow cover and freezing have also been shown to reduce radon emissions.

Example: In Northern Saskatchewan, Canada, mineralized waste rock has been placed in former open pit mines; several metres of water cover provide an effective barrier to oxygen and radon.

7.2.2.2. Earthen cover materials

Low permeability soil covers are generally made from fine grained silts and clays. These materials can achieve very low permeabilities of less than 1×10^{-9} m/s [102] and are very effective for limiting radon transport. However, permeability may increase for unprotected clay covers as a result of wetting and drying, freezing and thawing, and of deformation processes [103], which can lead to increased radon transport. A conventional cover system designed to limit infiltration and radon emissions has a lateral drainage layer consisting of either coarse sand or gravel, to limit head buildup on top of the clay, and a top layer of durable rock for erosion protection. Depending on the climate and environment, this rock may also support a vegetative cover.

Use of rock mulch and vegetation and avoidance of compaction may be strategies used for limiting erosion by rainfall [104], although using vegetation and avoiding its compaction may mean higher radon flux densities than would otherwise be the case. In such cases, the design must be optimized, taking into account the competing requirements and the specific characteristics of the situation.

Native soil covers may provide adequate resistance to wind and water erosion, and may also encourage indigenous plant species; for long term design, weather resistant natural materials are favoured. However, even if compacted, sandy native soils are much less effective in reducing radon emissions than other kinds of native soil, unless a considerable soil depth is used.

As discussed in Section 2.3, soil moisture can significantly affect radon transport. Moist and saturated soil conditions in cover materials tend to slow the rate of diffusion to the surface. The diffusion coefficient through a saturated soil may be several orders of magnitude smaller compared with that of a dry soil (Fig. 9).

Example: The decommissioning of the Cluff Lake Uranium Mine site in Canada involved placing a 1 m thick layer of locally available glacial till (the till was classified as a silty sand material with grading lying within the following envelope: 12–30% gravel, 58–64% sand, 16% silt and clay, 0–8% boulders and cobbles) on top of a residue impoundment containing 2.6 Mm³ of residues covering an area of 55 ha. After placement of the cover, radon concentration measurements were equal to background levels.

7.2.2.3. Geosynthetics

Manufactured from synthetic polymeric materials, geosynthetics are increasingly being used as covers in containment facilities, such as low level radioactive waste landfills. The two main types of low permeability geosynthetics used in cover design are geomembranes and geosynthetic clay liners. Other types of geosynthetics such as geonets and geotextiles are used to serve other functions in a cover system (e.g. used as filters to separate fines) and help to protect low permeability materials.

Geomembranes composed of high density polyethylene are most commonly used for lining residue facilities, while the more flexible polymers, such as low density polyethylene, polypropylene and polyvinyl chloride, are often used in final covers to accommodate settlement. These membrane materials have a very low hydraulic conductivity to water (less than 10⁻¹³ m/s); however, leakage can occur through defects such as tears or inadequate seams in the geomembrane. An assessment of leakage through holes in the geomembrane requires an estimate of the size and frequency of such holes. Ref. [105] provides guidance on estimating defects per hectare and Ref. [106] reviews gas migration through geomembranes. Long term covers should consider the degradation concerns for geomembranes. The estimated service life of a geomembrane can vary widely and is strongly dependent on the temperatures to which the geomembrane is exposed, ranging from about 1000 years at 10°C to as little as 15 years at 60°C. For this reason, geomembranes are often part of composite cover systems, where a second layer (of clay, for example) limits releases from membrane imperfections and thus extends the life of the cover system.

Geosynthetic clay liners (GCLs) consist of a layer of bentonite supported by one or two layers of fabric material. The bentonite layer is about 6 mm thick, and has a mass per unit area of $3.6-4.3 \text{ kg/m}^2$ and a water content of 10-20% at the time of manufacture. Although GCLs have a very low hydraulic conductivity, they may not be thick enough to provide sufficient time for radon to decay; GCLs are therefore also more effective when combined with other layers.

Example: The cover for a 32 ha low level radioactive waste landfill in Fernald, Ohio (1.0 Mm³ of mixed industrial, construction and low level radioactive waste), consists of a 2.9 m thick composite final cover system. The cover includes a geomembrane underlain by a geosynthetic clay liner and a compacted clay layer. The 0.9 m biointrusion layer placed on the surface of the cap system is designed to further limit the release of radon [103].

7.2.2.4. Evapotranspirative cover layers

Evapotranspirative cover layers rely on seasonal removal by evaporation and transpiration to control infiltrating surface water. They are becoming more common in arid or semiarid regions where potential evapotranspiration significantly exceeds actual precipitation [107]. Capillary barriers can be placed below the evapotranspirative cover. This would create a storage reservoir to enhance the available water capacity and limit unsaturated flow [108]. Available water storage capacity has been defined as the difference between the total amount of water remaining when the soil dries and the permanent wilting point for plants. A conservative choice would be to use the wilting point to determine how dry conditions predict the maximum exhalation of radon gas from a waste facility.

7.2.3. Effects of vegetation on the cover

Surface vegetation can be used to protect cover materials; it is used for erosion control, stabilization and limiting infiltration. However, investigations of cover systems have identified several mechanisms by which surface vegetation could compromise design criteria for mitigating radon release by cover materials:

- (a) Plants may act to alter soil moisture conditions, which can affect radon flux density. For example, plants may remove moisture from the soil during transpiration, which would result in a drier soil, thus increasing the radon flux density.
- (b) Root intrusion can increase the permeability of the underlying soil layers, resulting in an increased radon flux density. Roots from woody plants and deep shrubs have been observed to penetrate the compacted clay layer at several uranium mine waste sites [109].
- (c) Plants that are rooted deeper within the cover and waste system may act as conduits for radon. Ref. [110] shows that aqueous radon and radium are readily extracted from the soil into vascular plants via the transpiration stream and that radon is eventually released into the atmosphere through the leaf cuticle.

Numerical models have been developed to simulate radon and radium transport in a multiphase soil system that may or may not include vegetation. For more information, see Ref. [111].

8. MONITORING RADON RELEASES FROM NORM FACILITIES

8.1. GENERAL

Monitoring for radionuclides is undertaken at NORM facilities to demonstrate that adequate measures have been taken to protect the environment and keep radiological doses as low as reasonably achievable (ALARA). Many monitoring programmes are required by regulations, but the collected data are also beneficial for other reasons, such as for effective control of discharges and for public information.

Comprehensive information for the design and operation of source and environmental monitoring programmes and systems relating to the release of radioactive material to the environment from authorized practices can be found in Safety Reports Series No. 64 [112]. This section focuses on monitoring radon exhalation from NORM residue repositories.

A radon exhalation monitoring programme consists of a set of activities implemented to sample, measure and analyse radon in and around the area of a residue repository for the purpose of, but not limited to:

- Verifying compliance with any authorized limits or safety standards;
- Determining the source term for predictive modelling of radon levels;
- Obtaining data for the design and assessment of the performance of mitigation strategies such as waste covers.

The purpose, design and implementation of a radon monitoring programme are dependent on the facility's characteristics, environmental characteristics and anticipated effects. In particular, the monitoring programme will change according to the operational phase of the residue repository.

A monitoring programme should be planned in advance in order to ensure proper progression of the activity and the appropriate selection, use and optimization of resources.

Planning should define:

- The purpose of the monitoring plan;
- The design of the programme;
- Monitoring techniques and methods;
- Survey implementation;
- Data handling, analysis and maintenance;
- Calibration, quality control and quality assurance.

Sections 5 and 6 discussed the methods and tools available for calculating and measuring radon releases from NORM residues. This section briefly reviews the factors that should be considered in setting up a radon exhalation monitoring programme. The guidance given here is general in nature and relates only to the monitoring part of the overall monitoring programme. Since the design and implementation are site specific, expert advice on carrying out such activities should be obtained, especially if the information is to be used to design covers for residue repositories or to assess health effects to workers or nearby residents as well as to non-human biota.

8.2. DATA COLLECTION AT VARIOUS STAGES

Knowledge of the radon exhalation flux densities of a site prior to operation is a very important factor in assessing the above background radiological impact of a NORM residue repository during and after operation. Collection of suitable data prior to site operation is crucial as such data cannot be collected once the site is disturbed. For example, in the case of one uranium mining operation in northern Australia, it is possible that mining operation reduced the radon exhalation flux from the site because the original ore body outcropped at the surface, but the small pre-mining dataset does not enable the determination of a sound value for the true pre-mining exhalation flux [113].

Radon monitoring is conducted during the operational stage for periodic impact assessments along with the assessment of incremental changes in exhalation flux densities. These data will be used to guide decommissioning activities.

Post-operational monitoring is used to evaluate the adequacy of decommissioning activities, such as the ability of cover materials to limit radon exhalation.

Periodic and systematic review of the monitoring programme is necessary to ensure that the information being collected is still relevant and is fulfilling required objectives.

8.3. FACTORS IN DESIGN AND IMPLEMENTATION

A number of factors need to be considered in setting up a programme to assess the releases of radon from residue repositories. Many of these factors are site specific. To be effective, the monitoring programme should consider the following [112, 114]:

- Location and number of sampling stations;
- Location and number of control sites;
- Timing and frequency of measurements, taking into account diurnal or seasonal variations;
- Regulatory requirements for the accuracy of instruments and the design of the programme;
- Resource considerations;
- Physical, hydrogeological and radiological characteristics of the residue repository;
- Effect of meteorological parameters, such as precipitation, wind speed and direction, temperature, relative humidity and barometric pressure;

 Seasonal and ageing effects, such as changes in moisture content of the residue and the cover.

8.4. QUALITY ASSURANCE

Incorporating quality assurance activities into each aspect of the radon monitoring programme (project design, sampling, sample handling, data interpretation) will provide confidence in the results of the programme. The monitoring programme should address the following issues: instrument calibration and maintenance, the establishment of sampling and measurement protocols, adherence to relevant standards and best practices, and quality control.

Development of standardized documentation for recording and reporting results will facilitate the availability and comparability of data and documentation over a long period, which is normally required to track the performance of a residue repository.

9. CASE STUDY OF RADON EXHALATION FROM A URANIUM RESIDUE REPOSITORY

9.1. INTRODUCTION

This section presents a case study of multiparametric measurements carried out on a uranium waste repository at Jaduguda, India. The residues produced at the uranium mill at Jaduguda are segregated into coarse and fine fractions at the ore processing facility before disposal. The coarse fraction is used for backfilling the mine, while the fine fraction (slurry form) is placed into a tailings pond. The residues consolidate and form a compact mass of low permeability. These conditions favoured the use of a diffusion based model for predicting radon releases.

This study illuminates some of the results presented in the previous sections, by comparing predicted flux by means of empirical equations to direct measurements using an accumulator.

The radon flux densities were predicted using two approaches, namely:

(1) Using in situ measured values for the radon emanation coefficient (*E*) and radon diffusion length in the residues;

(2) Using the empirical correlations given for the radon diffusion coefficient $(D_r \text{ (m}^2/\text{s}))$ in Ref. [51] and for the radon emanation coefficient (*E*) in Ref. [68].

9.2. MEASURING RADON FLUX DIRECTLY AT THE RESIDUE SURFACE

In situ radon fluxes from the residue surface were measured using the accumulator technique. The set-up (Fig. 18) consisted of a cylindrical accumulator (15 cm in diameter and 15 cm in height) attached to a continuous radon monitor.



FIG. 18. Accumulator with AlphaGuard continuous radon monitor set up for radon flux measurement.

The effective volume of this measurement system is 3.4 L. The accumulator was inserted into the residue surface to a depth of approximately 1 cm to prevent superficial leakage. Radon concentrations inside the accumulator were monitored at 10 min time intervals for about 3 hours at each location. The concentration data were fitted to an exponential growth model to simultaneously obtain the in situ radon flux and diffusion coefficient by following the methods outlined in Sections 3.2 and 5.3.1.

Typical data on the buildup of radon concentration in the accumulator deployed at one location at the site are shown in Fig. 19.



FIG. 19. Typical buildup data of radon concentration in the chamber with least square fitting of Eq. (16) in Section 5.3.1.2. for an accumulator of radius a = 7.5 cm and effective volume V = 3.4 L deployed on the uranium residue repository.

9.3. ESTIMATING FLUX BY MEASUREMENT OF L, AND E

Residue samples were collected from 40 locations of the residue pile by inserting a 1000 cm³ stainless steel cylindrical tube into the residue matrix. The samples were processed and dried in an oven at 110°C to attain constant weight. Then, the samples were packed in a 300 cm³ leakproof plastic container and stored for approximately 30 d to ensure secular equilibrium between ²²⁶Ra and its decay products. Gamma spectrometry [115] was used to measure the dry weight specific ²²⁶Ra content in the samples. The samples were subjected to radon mass exhalation rate measurement [94] as outlined in Section 5.3.5. The dry sample radon emanation coefficient was then determined by the equation:

$$E = J_{\rm m} / \lambda R \tag{28}$$

where

E is the dry sample radon emanation coefficient;

 J_m is the mass exhalation rate (Bq·kg⁻¹·s⁻¹);

R is the dry weight radium activity concentration (Bq/kg);

and λ is the decay constant of ²²²Rn (s⁻¹).

The soil porosity was estimated using the equation:

$$n_T = 1 - \rho_b / \rho_g \tag{29}$$

where

 n_T is the soil porosity; ρ_b is the measured dry bulk density of the residue (kg/m³);

and ρ_{o} is the grain density of the residue material (2700 kg/m³) [116].

The moisture saturation fraction (*m*), the fraction of the pore space filled with water, was estimated from the measured soil moisture as $m = \rho_b \theta_d / 100 \rho_w n_T$ where θ_d is the moisture content on a dry weight basis and ρ_w is the density of water (kg/m³). The in situ temperature *T* at a depth of 10 cm from the residue surface was measured by a sensor.

9.3.1. In situ radon diffusion length (L_r)

The in situ radon diffusion lengths were determined by analysing the depth profile of radon concentration in the residue matrix. The schematic diagram of the measurement set-up is shown in Fig. 20.

Radon concentrations were measured at different depths from 20 to 150 cm, using a soil probe with a flow rate of approximately 0.5 L/min. According to diffusion theory, the radon concentration C(z) at depth z follows the equation [19]:

$$C(z) = C_{\infty} \left(1 - \exp(-z/L_r) \right)$$
(30)

where

 C_{∞} is the radon concentration at great depth (Bq/m³);

z is the thickness of the material (m);

and L_r is the radon diffusion length (m).



FIG. 20. Schematic diagram of soil probe method for the measurement of radon concentration in residues using a continuous radon monitor.

At the end of the experiment, the radon concentration data at different depths were fitted to Eq. (30) by the method of least squares and the in situ radon diffusion length was obtained from the fitting coefficients. A typical fit to the data is shown in Fig. 21.



FIG. 21. Depth profile of radon concentration in soil gas and the fitted curve.

9.3.2. In situ radon emanation coefficient (E)

This process was performed along the lines outlined in Section 3.1. The in situ radon emanation coefficient (E) in the residue was determined by measuring the mass exhalation rates using fresh samples without subjecting them to drying. These experiments were performed at the site itself immediately after collecting the samples and hence the results are expected to be close to the values corresponding to in situ moisture conditions.

9.3.3. Estimation

According to Ref. [116], the radon flux at the residue surface may be written as:

$$f_r = \lambda L_r R \rho_b E \tag{31}$$

where

 f_r is the radon flux (Bq·m⁻²·s⁻¹);

 $\dot{\lambda}$ is the decay constant of ²²²Rn (s⁻¹);

R is the radium activity concentration in uranium residue (Bq/kg);

and ρ_{h} is the bulk density of residues (kg/m³).

9.4. ESTIMATING FLUX BY CALCULATING L_r AND EVIA EMPIRICAL FORMULAS

The third, and least exact, approach to estimating the radon flux at the surface uses Eq. (31), but, instead of using experimentally measured values of L_r (and thus D_r) and E, it uses empirical relations for D_r and E. The diffusion coefficient was estimated using Eq. (14), including a temperature correction:

$$D_r = D_{MA} n_T \exp(-6mn_T - 6m^{14n_T}) \left(\frac{T}{273}\right)^{0.75}$$
(32)

The following empirical equation was employed to estimate *E*:

$$E = E_0 \Big[1 + 1.85 \big(1 - \exp(-18.8m) \big) \Big]$$
(33)

59

The variables in the above equations are defined as follows:

- D_{MA} is the radon diffusion coefficient in air at ambient temperature and pressure $(1.1 \times 10^{-5} \text{ m}^2/\text{s});$
- n_{τ} is the soil porosity;
- m is the fraction of pore space filled with water;
- *T* is the absolute temperature (K);

and E_0 is the radon emanation coefficient for uranium residue in dry conditions.

Once these equations were computed, they were simply substituted into Eq. (29) to generate the flux estimate.

9.5. ILLUSTRATIONS OF ESTIMATION AND MEASUREMENT RESULTS

9.5.1. Determination of exhalation flux

Table 5 presents a summary of various parameters measured for the repository. Figure 22 shows the radon fluxes measured and predicted (by the two approaches mentioned above) over 10 locations of the residue. There is an overall agreement, within deviations of approximately 20%, between the measured and the predicted radon fluxes. A closer statistical analysis (not shown) indicates that the radon fluxes predicted using the in situ diffusion lengths and emanation coefficients are closer to the measured values than the fluxes predicted from the moisture corrected empirical functions given in Eqs (32) and (33). This could be due to the fact that the moisture content estimated from the top layer of the repository may not represent values down to a depth of approximately 1 m, thereby affecting the parameter values based on empirical correlations.

9.5.2. Determination of radon diffusion length

Sometimes, it is of interest to estimate the diffusion length, given a known exhalation flux. For this purpose, the relevant equations can be reversed and the known values of exhalation flux can be used to estimate Lr using measurement based and empirical formulas.

TABLE 5. THE MEAN OF MEASURED RESIDUE PARAMETERS INCLUDING STANDARD DEVIATION IN WINTER, SUMMER AND RAINY SEASONS FOR 40 LOCATIONS OF THE URANIUM WASTE RESIDUE REPOSITORY

Parameters	Winter	Summer	Rainy	Average
Radium, <i>R</i> (Bq/kg)	5124 ± 2348	5144 ± 2256	5081 ± 2244	5166 ± 2282
In situ ²²² Rn diffusion length, L_r (m)	0.81 ± 0.15	0.90 ± 0.14	0.72 ± 0.14	0.81 ± 0.14
In situ ²²² Rn emanation coefficient, E	0.24 ± 0.05	0.20 ± 0.02	0.28 ± 0.07	0.24 ± 0.05
Dry ²²² Rn emanation coefficient, E_0	0.09 ± 0.04	0.10 ± 0.03	0.09 ± 0.03	0.09 ± 0.03
Dry bulk density, $\rho_b(kg/m^3)$	1790 ± 66	1786 ± 52	1801 ± 37	1792 ± 52
Total porosity, $n_{\rm T}$	0.34 ± 0.02	0.34 ± 0.03	0.33 ± 0.04	0.34 ± 0.03
Moisture saturation fraction, <i>m</i>	0.54 ± 0.05	0.35 ± 0.04	0.74 ± 0.06	0.54 ± 0.05



FIG. 22. Comparison between measured and predicted radon flux densities at 10 locations.
Figure 23 shows the comparison of radon diffusion lengths obtained from the three methods: the accumulator method, the depth profile analysis of soil gas radon and the empirical relation given in Eq. (32). Overall agreement between the three methods was within 20%. The diffusion lengths estimated from the empirical correlation method were generally higher than the other two, thereby supporting the moisture content effect mentioned in Section 9.5.1. The study also demonstrates that the accumulator method is a useful tool, not only for deriving exhalation flux, but also for estimating the in situ diffusion coefficient in soil.



FIG. 23. Comparison of radon diffusion length in uranium residue matrix obtained by three different methods at 10 locations.

9.5.3. Relating radium content to radon flux

Separately from the above analysis, Fig. 24 shows the relationship between radium content and the seasonally averaged radon fluxes at over 40 locations on the repository. The linear fit shows a strong correlation between radium content and radon flux in the repository. The slope yields a ratio of $(8.3 \pm 0.4) \times 10^{-4} \text{ kg} \cdot \text{m}^{-2} \cdot \text{s}^{-1}$. This ratio also represents the product of the radon

decay constant λ (2.1 × 10⁻⁶) and average values of the radon diffusion length (L_r), the soil density (ρ_b) and the in situ radon emanation coefficient (*E*). Using the measured value of the above parameters (Table 5), this ratio works out to be 7.3 × 10⁻⁴ kg·m⁻²·s⁻¹. The slope (8.3 ± 0.4) × 10⁻⁴ kg·m⁻²·s⁻¹ may be used as a conversion factor to quickly estimate the radon flux (Bq·m⁻²·s⁻¹) using the ²²⁶Ra content (Bq/kg) for the uranium waste residue repository at Jaduguda.



FIG. 24. Correlation between measured radon flux densities and ²²⁶Ra content at 40 locations.

Appendix I

MATHEMATICAL DEVELOPMENT OF RADON DIFFUSION EQUATIONS

I.1. FORMULATION OF RADON DIFFUSION EQUATION FOR A POROUS MEDIUM

Consider a matrix in which radon is continuously released to the pore volume of the matrix owing to emanation from the grains containing Ra. Let P_p be the radon activity released into the unit volume of the pore space per unit time. A local flux density f_p (a vector quantity) may be defined representing the activity of radon crossing per unit pore area per unit time. Let C_p be the pore space radon concentration (i.e. radon activity divided by volume of pore space) at any point at a given time. The general transport equation is obtained by a limiting process of the time rate of change of radon activity in an infinitesimal pore volume as being due to the difference between the generation rate and losses due to leakage rate and radioactive decay (with decay constant λ) [19]:

$$\frac{\partial C_p}{\partial t} = P_p - \nabla \cdot f_p - \lambda C_p \tag{34}$$

In general, f_p has two components: (a) pressure driven advection flux and (b) gradient driven diffusion flux. Generally, in most situations, the pressure difference sustained between the pore air and the atmosphere is too small to cause significant pressure induced flux as compared with diffusion driven flux and, hence, it is neglected. For a one dimensional system, the diffusion flux is expressed in terms of the concentration gradient by applying Fick's first law of diffusion as follows:

$$f_p = -D\frac{\partial C_p}{\partial z} \tag{35}$$

where *D* is the radon diffusion coefficient in the pore space in the matrix. In a completely dry material, *D* is the product of the molecular diffusion coefficient (D_{MA}) in the pore space air and tortuosity (τ) that accounts for the tortuous paths traversed by the radon atoms along the pores [117]. However, in general, residues contain some moisture and thus radon atoms will diffuse both in the air and water phases resulting in a lower diffusion coefficient (*D*) in the pore space [51].

Substitution of Eq. (35) into Eq. (34) yields the following diffusion equation:

$$\frac{\partial C_p}{\partial t} = P_p + D \frac{\partial^2 C_p}{\partial z^2} - \lambda C_p \tag{36}$$

If *R* is the radium activity concentration of the matrix (i.e. the radium activity in the soil grain divided by the mass of the matrix), ρ_b is the bulk density of the matrix (i.e. the mass of the matrix divided by the bulk volume of the matrix), *E* is the radon emanation coefficient and n_T is the total porosity of the matrix (i.e. the total pore volume divided by the bulk volume), then P_p can be expressed as:

$$P_p = \frac{\lambda R \rho_b E}{n_T} \tag{37}$$

With this, Eq. (36) becomes:

$$\frac{\partial C_p}{\partial t} = D \frac{\partial^2 C_p}{\partial z^2} - \lambda C_p + \frac{\lambda R \rho_b E}{n_T}$$
(38)

While the flux f_p in Eq. (35) refers to unit pore area, the exhalation flux at the residue surface refers to the bulk area that includes both the pore area and the area covered by the solid materials. Assuming the fractional pore area at the surface is the same as the fractional pore volume of the bulk of the residue, the exhalation flux density may be related to the pore space concentration as follows:

$$f = n_T f_p = -n_T D \frac{\partial C_p}{\partial z}$$
(39)

In dry residues, the pores will be filled only with air and, hence, C_p will be the same as the concentration C_a in pore air. In the case of residues containing moisture, the radon atoms in the pore volume will be distributed partly in air and partly in water in proportion to the water–air partition coefficient of radon. It is more convenient to recast the diffusion equation in terms of C_a to develop consistent formulations and interface boundary conditions to address multilayer problems involving covers of different moisture contents and material properties. If C_w and C_a denote the radon concentration in water and in air, and if n_w and n_a are water filled and air filled porosities respectively, then mass conservation [117] requires that:

$$n_T C_p = n_a C_a + n_w C_w \tag{40}$$

The law of equilibrium partitioning relates C_w to C_a as follows:

$$C_w = KC_a \tag{41}$$

where *K* is the water–air partition coefficient. Combining Eq. (41) with Eq. (40) gives:

$$n_T C_p = C_a (n_a + K n_w) = n_e C_a \tag{42}$$

where

$$n_e = (n_a + K n_w) \tag{43}$$

The quantity n_e denotes the partition corrected porosity. It may be expressed more conveniently in terms of the volumetric moisture saturation (*m*), defined as $m = n_w/n_{\tau^2}$ as follows :

$$n_e = n_T [1 - (1 - K)m] \tag{44}$$

With this, Eq. (38) may be recast in terms of C_a as follows:

$$\frac{\partial C_a}{\partial t} = D \frac{\partial^2 C_a}{\partial z^2} - \lambda C_a + \frac{\lambda R \rho_b E}{n_e}$$
(45)

Under steady state conditions, Eq. (45) becomes:

$$D\frac{\partial^2 C_a}{\partial z^2} - \lambda C_a + \frac{\lambda R \rho_b E}{n_e} = 0$$
(46)

The corresponding expression for flux density is:

$$f = -n_e D \frac{\partial C_a}{\partial z} \tag{47}$$

67

The formulation of the diffusion equation in terms of radon concentration in the air phase (C_a) and partition corrected porosity (n_e) enables a direct application to a multilayer problem since both C_a and f will satisfy continuity requirements at the interface of the two layers even if their moisture content and porosities differ from one other.

I.2. GENERAL SOLUTION AND BOUNDARY CONDITIONS

The general solution to Eq. (46) is:

$$C_a(z) = A \exp(z/L) + B \exp(-z/L) + (R\rho_b E/n_e)$$

$$\tag{48}$$

where

$$L = \sqrt{\frac{D}{\lambda}} \tag{49}$$

and is known as the diffusion length of radon in the given matrix. A and B are unknown coefficients to be determined using the boundary conditions of the problem.

The associated radon flux density, obtained from Eqs (47) and (48), is:

$$f = \frac{n_e D}{L} \Big[B \exp(-z/L) - A \exp(z/L) \Big]$$
(50)

For simplicity of notation in multilayer problems, it is convenient in Eq. (48) to drop the subscript 'a' denoting the radon concentrations in the air of the pore space and replace it with layer (i = 1, 2, 3...) specific radon concentrations. Hence, the radon concentration in the air space of the *i*th layer will be denoted by C_i . Similarly, the subscript 'e' may be dropped from the partition corrected porosity since this is the only porosity that actually appears in all the expressions. With this, the boundary conditions of the problem may be stated as:

(a) f(0) = 0 the flux is zero at the base of the residues (impervious base);
(b) C_i(z_i) = C_{i+1}(z_i) the concentration is continuous across the interface at z_i;
(c) f_i(z_i) = f_{i+1}(z_i) the flux is continuous across the interface at z_i;

(d) $C_n(z_n) = C_{atm}$ the concentration at the surface of the top layer (medium *n*) is equal to a specified value, usually the atmospheric concentration.

I.3. SOLUTION FOR RADON FLUX ACROSS THE SURFACE OF BARE RESIDUES

The application of boundary condition (a) to Eq. (50) yields A = B. The application of boundary condition (d) for $C_{atm} = 0$ at the top surface of the residues of thickness z_{x} yields values for A as a function of z_{x} .

Finally, the flux at the surface of the bare residues is given by:

$$f(z_r) = f_r = R\rho E L_r \lambda \tanh\left[\frac{z_r}{L_r}\right]$$
(51)

where subscript 'r' refers to residues.

I.4. SOLUTION FOR COVERED RESIDUES

The solution of the diffusion equation for a two region problem applies to a residue pile covered with a homogeneous material. For simplicity, the source term in the cover is assumed to be zero, and C_{atm} at the surface of the cover is also assumed to be zero.

Application of the boundary conditions yields values for A_c and B_c as functions of z_c , where the subscript 'c' refers to the cover material. Substitution of A_c and B_c into Eq. (50) yields the flux as a function of the position z in the cover. Finally, the flux at the surface of the cover is given by substituting $z = z_c$:

$$f = \frac{2f_r \exp\left[-\frac{z_c}{L_c}\right]}{\left[1 + \frac{n_r L_r}{n_c L_c} \tanh\frac{z_r}{L_r}\right] + \left[1 - \frac{n_r L_r}{n_c L_c} \tanh\frac{z_r}{L_r}\right] \exp\left[-2\frac{z_c}{L_c}\right]}$$
(52)

The terms n_i and L_i account for moisture, porosity and diffusion coefficient differences at the residue–cover interface. If the moisture and porosity differences are small, the residue and cover diffusion coefficients are approximately equal and residue thickness is large compared with z_{c2} then Eq. (52) becomes:

$$f_c \approx f_r \exp\left[-\frac{z_c}{L_c}\right] \tag{53}$$

This implies that the free surface flux decreases exponentially as the thickness of the cover increases. This relationship is useful for a quick estimation of the cover thickness required to achieve a given exhalation flux density.

Appendix II

REFINED METHODS OF RADON FLUX CALCULATION

The equations in Section 6 (simplified expressions for estimating radon exhalation flux density) can be modified to increase the accuracy of the calculations and extend their range of application. This is shown here for the cases of uncovered and single layer systems. Exact analytical solutions for up to four layer systems have been developed [118–120], but they are cumbersome to use. In general, multilayer systems are addressed by solving the differential equations directly using numerical models.

II.1. FLUX CALCULATION FOR UNCOVERED LAYERED RESIDUES

Some residue situations require that the vertical heterogeneities in a residue pile be considered explicitly. Such heterogeneity may be due to major variations in the radium concentration or in the diffusion coefficient as a function of depth. The heterogeneities are considered explicitly by dividing them into layers that have homogeneous properties within each layer. Then, each layer is both a source layer and a cover layer to layers beneath it. Combining Eqs (51) and (53) gives, for the flux at the surface of a pile with n layers:

$$f_r = \sum_{i=1}^n R_i \rho_i E_i \lambda L_{ri} \tanh\left[\frac{z_{ri}}{L_{ri}}\right] \exp\left[-\sum_{j=i+1}^n \frac{z_{ri}}{L_{ri}}\right]$$
(54)

where

$$L_{ri} = \sqrt{\frac{D_{ri}}{\lambda}} \tag{55}$$

is the diffusion length of radon in the *i*th layer of the residue material.

In the formula given above, the following assumptions have been made:

- (a) For a given layer, the layer beneath it is treated as impervious to radon;
- (b) The approximation given in Eq. (53) is used to represent the effect of a cover layer above the given residue layer.

For example, consider a three layer residue pile. The properties of the pile are described in Table 6. The layers are heterogeneous both in depth and in diffusion coefficient (D_r) . An exact calculation performed with the RAECOM [120] computer code yields a randon flux density (f_r) of 17 Bq·m⁻²·s⁻¹. Using these same variables in Eq. (54) yields an f_r of 18 Bq·m⁻²·s⁻¹. However, if the D_r were deemed homogeneous throughout the three layers using the D_r of the middle layer, the resulting f_r would be 12 Bq·m⁻²·s⁻¹. If the D_r were deemed homogeneous with the top layer value, the resulting f_r would be 17 Bq·m⁻²·s⁻¹. This highlights the higher influence of the properties of the top layer of residue on the f_r of the pile.

Layer	Layer depth (m)	Diffusion coefficient (D_r)	Radium activity concentration (Bq/kg)	Emanation coefficent (E)	Density (kg/m ³) (ρ)	Radon flux density (f_r)
All three layers	6		4×10^4	0.2	1.5×10^{3}	
Тор	1	$1\times 10^{-6}\ m^2/s$				18 Bq \cdot m ⁻² \cdot s ⁻¹
Middle	2	$5\times 10^{-7}\ m^2/s$				
Bottom	3	$1\times 10^{-7}\ m^2/s$				
Layers homogeneous with middle D_r	6	$5 imes 10^{-7} \text{ m}^2/\text{s}$				$12 \text{ Bq} \cdot m^{-2} \cdot s^{-1}$
Layers homogeneous with top D_r	6	$1 \times 10^{-6} \text{ m}^2/\text{s}$				17 Bq·m ⁻² ·s ⁻¹

TABLE 6. PROPERTIES OF AN EXAMPLE THREE LAYER RESIDUE PILE

For calculating ²²⁰Rn fluxes from thorium residues, the properties of the top layer (a few centimetres deep) are sufficient.

II.2. FLUX CALCULATION FOR A SINGLE COVER

Apart from the thickness of the cover material, the properties of the cover and the residue matrices have an important bearing on the radon flux reduction factor. This is illustrated below. The exact solution of the radon diffusion equation for a single cover in Eq. (52) is:

$$f_{c} = \frac{2f_{r} \exp\left[-\frac{z_{c}}{L_{c}}\right]}{\left[1 + \frac{n_{r}L_{r}}{n_{c}L_{c}} \tanh\frac{z_{r}}{L_{r}}\right] + \left[1 - \frac{n_{r}L_{r}}{n_{c}L_{c}} \tanh\frac{z_{r}}{L_{r}}\right] \exp\left[-2\frac{z_{c}}{L_{c}}\right]}$$
(56)

where

$$n_i = n_{Ti} [1 - (1 - K)m_i], \ i = r \text{ or } c$$
(57)

denotes the partition corrected porosity for residue r or cover c, and

$$L_i = \sqrt{\frac{D_i}{\lambda}, \, i = r \, \text{or} \, c} \tag{58}$$

denotes the radon diffusion length in residue r or cover c.

K denotes the radon distribution coefficient between water and air (about 0.26 at 20°C), and m_i denotes the volume fractional moisture saturation in residues or cover.

If the residue pile thickness is much greater than the diffusion length of radon, then the hyperbolic tangent term is unity and Eq. (56) becomes:

$$f_{c} = \frac{2f_{r} \exp\left[-\frac{z_{c}}{L_{c}}\right]}{1 + \frac{n_{r}L_{r}}{n_{c}L_{c}} + \left[1 - \frac{n_{r}L_{r}}{n_{c}L_{c}}\right] \exp\left[-2\frac{z_{c}}{L_{c}}\right]}, z_{r} >> z_{c}$$
(59.a)

If the moisture and porosity differences between the residue and cover layers are small and if corresponding diffusion coefficients are approximately equal, Eq. (59.a) further reduces to:

$$f_c = f_r \exp\left[-\frac{z_c}{L_c}\right]$$
(59.b)

which is the simple exponential attenuation shown by A in Fig. 25.

However, if $n_r L_r$ is much less than $n_c L_c$, Eq. (56) becomes:

$$f_c = \frac{2f_r \exp\left[-\frac{z_c}{L_c}\right]}{1 + \exp\left[-2\frac{z_c}{L_c}\right]}$$
(60)

If the value of z_c is low, the value of f_c is approximately equal to f_r and very little flux attenuation occurs, as shown by B in Fig. 25. However, as z_c increases, f_c decreases exponentially in the same manner as in A, but retains twice the magnitude.

The case in which $n_r L_r$ is much greater than $n_c L_c$ is shown as C in Fig. 25. In this case, a thin cover is relatively effective; f_c again decreases, but its magnitude is $2(n_c L_c/n_r L_r)$ times the value in curve A. However, in most situations, it is very difficult to maintain a cover with an $n_c L_c$ value less than the $n_r L_r$ value of the residue. Hence, this condition has limited application.



FIG. 25. Radon flux reduction factor for various cover thicknesses. (A: $n_r L_r = n_c L_c$: B: $n_r L_r < < n_c L_c$; C: $n_r L_r >> n_c L_c$).

REFERENCES

- OECD NUCLEAR ENERGY AGENCY, INTERNATIONAL ATOMIC ENERGY AGENCY, Uranium 2009: Resources, Production and Demand, OECD Publishing, Paris (2010).
- [2] INTERNATIONAL ATOMIC ENERGY AGENCY, Measurement and Calculation of Radon Releases from Uranium Mill Tailings, Technical Reports Series No. 333, IAEA, Vienna (1992).
- [3] INTERNATIONAL ATOMIC ENERGY AGENCY, Characterization of Radioactively Contaminated Sites for Remediation Purposes, IAEA-TECDOC-1017, IAEA, Vienna (1998).
- [4] INTERNATIONAL ATOMIC ENERGY AGENCY, Factors for Formulating Strategies for Environmental Restoration, IAEA-TECDOC-1032, IAEA, Vienna (1998).
- [5] INTERNATIONAL ATOMIC ENERGY AGENCY, Design Criteria for a Worldwide Directory of Radioactively Contaminated Sites (DRCS), IAEA-TECDOC-1251, IAEA, Vienna (2001).
- [6] INTERNATIONAL ATOMIC ENERGY AGENCY, Non-technical Factors Impacting on the Decision Making Processes in Environmental Remediation, IAEA-TECDOC-1279, IAEA, Vienna (2002).
- [7] INTERNATIONAL ATOMIC ENERGY AGENCY, Site Characterization Techniques Used in Environmental Restoration Activities, IAEA-TECDOC-1148, IAEA, Vienna (2000).
- [8] INTERNATIONAL ATOMIC ENERGY AGENCY, Extent of Environmental Contamination by Naturally Occurring Radioactive Material (NORM) and Technological Options for Mitigation, Technical Reports Series No. 419, IAEA, Vienna (2003).
- [9] INTERNATIONAL ATOMIC ENERGY AGENCY, Technologies for Remediation of Radioactively Contaminated Sites, IAEA-TECDOC-1086, IAEA, Vienna (1999).
- [10] INTERNATIONAL ATOMIC ENERGY AGENCY, Technical Options for the Remediation of Contaminated Groundwater, IAEA-TECDOC-1088, IAEA, Vienna (1999).
- [11] INTERNATIONAL ATOMIC ENERGY AGENCY, Compliance Monitoring for Remediated Sites, IAEA-TECDOC-1118, IAEA, Vienna (1999).
- [12] INTERNATIONAL ATOMIC ENERGY AGENCY, The Long Term Stabilization of Uranium Mill Tailings, IAEA-TECDOC-1403, IAEA, Vienna (2004).
- [13] INTERNATIONAL ATOMIC ENERGY AGENCY, Remediation of Sites with Dispersed Radioactive Contamination, Technical Reports Series No. 424, IAEA, Vienna (2004)
- [14] INTERNATIONAL ATOMIC ENERGY AGENCY, Radiation Protection and Safety of Radiation Sources: International Basic Safety Standards — Interim Edition, General Safety Requirements Part 3, IAEA Safety Standards Series No. GSR Part 3 (Interim), IAEA, Vienna (2011).

- [15] INTERNATIONAL ATOMIC ENERGY AGENCY, Radiation Protection and the Management of Radioactive Waste in the Oil and Gas Industry, Safety Reports Series No. 34, IAEA, Vienna (2003).
- [16] INTERNATIONAL ATOMIC ENERGY AGENCY, Generic Models for Use in Assessing the Impact of Discharges of Radioactive Substances to the Environment, Safety Reports Series No. 19, IAEA, Vienna (2001).
- [17] INTERNATIONAL ATOMIC ENERGY AGENCY, Regulatory Control of Radioactive Discharges to the Environment, IAEA Safety Standards Series No. WS-G-2.3, IAEA, Vienna (2000).
- [18] BE, M.-M., et al., Recommended Data, http://www.nucleide.org/DDEP_WG/DDEPdata.htm, accessed 2013-06-27.
- [19] MOED, B.A., NAZAROFF, W.W., SEXTRO, R.G., "Soil as a source of indoor radon: Generation, migration and entry", Radon and its Decay Products in Indoor Air (NAZAROFF, W.W., NERO Jr., A.V., Eds), John Wiley and Sons, New York (1988) 57–112.
- [20] TANNER, A.B., "Radon migration in the ground: A review", The Natural Radiation Environment (ADAMS, J.A.S., LOWDER, W.M., Eds), University of Chicago Press, Chicago, IL (1964).
- [21] SEMKOW, T.M., Fractal model of radon emanation from solids, Phys. Rev. Lett. 66 23 (1991) 3012–3015.
- [22] ZIEGLER, J.F., BIERSACK, J.P., LITTMARK, U., The Stopping and Range of Ions in Solids, Pergamon Press, New York (1985).
- [23] SAKODA, A., ISHIMORI, Y., YAMAOKA, K., A comprehensive review of radon emanation measurements for mineral, rock, soil, mill tailing and fly ash, Appl. Radiat. Isot. 69 (2011) 1422–1435.
- [24] FLÜGGE, S., ZIMENS, K.E., Die Bestimmung von Korngrössen und von Diffusionskonstanten aus dem Emaniervermögen (Die Theorie der Emaniermethode), Z. Phys. Chem. B42 (1939) 179–220.
- [25] FLEISCHER, R.L., Theory of alpha recoil effects on radon release and isotopic disequilibrium, Geochim. Cosmochim. Acta 47 (1983) 779–784.
- [26] MORAWSKA, L., PHILLIPS, C.R., Dependence of the radon emanation coefficient on radium distribution and internal structure of the material, Geochim. Cosmochim. Acta 57 (1993) 1783–1797.
- [27] SASAKI, T., GUNJI, Y., OKUDA, T., Radon emanation dependence on grain configuration, J. Nucl. Sci. Technol. 41 (2004) 993–1002.
- [28] BARILLON, R., OZUGUMUS, A., CHAMBAUDET, A., Direct recoil radon emanation from crystalline phases. Influence of moisture content, Geochim. Cosmochim. Acta 69 (2005) 2735–2744.
- [29] SAKODA, A., et al., First model of the effect of grain size on radon emanation, Appl. Radiat. Isot. 68 (2010) 1169–1172.
- [30] SAKODA, A., et al., Experimental and modeling studies of grain size and moisture content effects on radon emanation, Radiat. Meas. 45 (2010) 204–210.
- [31] KALKWARF, D.R., JACKSON, P.O., KUTT, J.C., Emanation coefficient for Rn in sized coal fly ash, Health Phys. 48 (1985) 429–436.

- [32] MARKKANEN, M., ARVELA, H., Radon emanation from soils, Radiat. Prot. Dosimetry 45 (1992) 269–272.
- [33] GARVER, E., BASKARAN, M., Effects of heating on the emanation rates of radon-222 from a suite of natural minerals, Appl. Radiat. Isot. 61 6 (2004)1477–1485.
- [34] KOVLER, K., PEREVALOV, A., STEINER, V., METZGER, L.A., Radon exhalation of cementitious materials made with coal fly ash: Part 1 — scientific background and testing of the cement and fly ash emanation, J. Environ. Radioact. 82 (2005) 321–334.
- [35] BREITNER, D., ARVELA, H., HELLMUTH, K., RENVALL, T., Effect of moisture content on emanation at different grain size fractions — A pilot study on granitic esker sand sample, J. Environ. Radioact. **101** (2010) 1002–1006.
- [36] BARTON, T.P., ZIEMER, P.L., The effects of particle size and moisture content on the emanation of Rn from coal ash, Health Phys. **50** (1986) 581–588.
- [37] STRONG, K.P., LEVINS, D.M., Effect of moisture content on radon emanation from uranium ore and tailings, Health Phys. 42 (1982) 27–32.
- [38] ROGERS, V.C., NIELSON, K.K., MERREL, G.B., Radon Attenuation with Earthen Covers, 1983 Annual Report of Rogers and Associates Engineering Corporation, Rep. RAE363, Salt Lake City, UT (1983).
- [39] BOSSEW, P., The radon emanation power of building materials, soils and rocks, Appl. Radiat. Isot. 59 (2003) 389–392.
- [40] SEMKOW, T.M., PAREKH, P.P., The role of radium distribution and porosity in radon emanation from solids, Geophys. Res. Lett. 17 (1990) 837–840.
- [41] LOTTERMOSER, B.G., ASHLEY, P.M., Tailings dam seepage at the rehabilitated Mary Kathleen uranium mine, Australia, J Geochem. Explor. 85 3 (2005) 119–137.
- [42] LANGMUIR, D., MAHONEY, J., McDONALD, A., ROWSON, J., Predicting arsenic concentrations in the porewaters of buried uranium mill tailings, Geochim. Cosmochim. Acta 63 19–20 (1999) 3379–3394.
- [43] BASSOT, S., STAMMOSE, D., BENITAH, S., Radium behaviour during ferric oxihydroxides ageing, Radioprotection, 40 S1 (2005) S277–S283.
- [44] MARTIN, A.J., CRUSIUS, J., McNEE, J.J., YANFUL, E.K., The mobility of radium-226 and trace metals in pre-oxidized subaqueous uranium mill tailings, Appl. Geochem. 18 7 (2003) 1095–1110.
- [45] RAMA, MOORE, W.S., Mechanism of transport of U-Th series radioisotopes from solids into ground water, Geochim. Cosmochim. Acta 48 (1984) 395–399.
- [46] RAMA, MOORE, W.S., Micro-crystallinity in radioactive minerals, Nucl. Geophys. 4 (1990) 475–478.
- [47] HOWARD, A.J., SIMSARIAN, J.E., STRANGE, W.P., Measurements of ²²⁰Rn emanation from rocks, Health Phys. 69 (1995) 936–943.
- [48] MEGUMI, K., MAMURO, T., Emanation and exhalation of radon and thoron gases from soil particles, J. Geophys. Res. 79 (1974) 3357–3360.
- [49] GREEMAN, D.J., ROSE, A.W., Factors controlling the emanation of radon and thoron in soils of the eastern U.S.A, Chem. Geol. 129 (1996) 1–14.
- [50] SCHUMANN, R.R., GUNDERSEN, L.C.S., Geologic and climatic controls on the radon emanation coefficient, Environ. Int. **22** S1 (1996) 439–446.

- [51] ROGERS, V.C., NIELSON, K.K., Correlations for predicting air permeabilities and ²²²Rn diffusion coefficients of soil, Health Phys. **61** 2 (1991) 225–230.
- [52] SCHROEDER, G.L., KRANER, H.W., EVANS, R.D., Diffusion of radon in several naturally occurring soil types, J. Geophys. Res. 70 2 (1965) 471–474.
- [53] PAPACHRISTODOULOU, C., IOANNIDES, K., SPATHIS, S., The effect of moisture content on radon diffusion through soil: assessment in laboratory and field experiments, Health Phys. 92 3 (2007) 257–264.
- [54] INTERNATIONAL ATOMIC ENERGY AGENCY, Analytical Methodology for the Determination of Radium Isotopes in Environmental Samples, IAEA Analytical Quality in Nuclear Applications Series No. 19, IAEA, Vienna (2010).
- [55] JONSSON, G., Soil radon depth dependence, Radiat. Meas. 34 1 (2001) 415–418.
- [56] WHITE, G.J., ROOD, A.S., Radon emanation from NORM-contaminated pipe scale and soil at petroleum industry sites, J. Environ. Radioact. 54 (2001) 401–413.
- [57] SAKODA, A., et al., Differences of natural radioactivity and radon emanation fraction among constituent minerals of rock or soil, Appl. Radiat. Isot. 68 (2010) 1180–1184.
- [58] SAHOO, B.K., et al., Radon exhalation studies in an Indian Uranium tailings pile, Radiat. Meas. 45 (2010) 237–241.
- [59] ROGERS, V.C., NIELSON, K.K., HOLT, R.B., SNODDY, R., Radon diffusion coefficients for residential concretes, Health Phys. 67 3 (1994) 261–265.
- [60] CHUNNAN, H.W., SHIHCHIN, T., SHIHMING, L., Evaluation of diffusion parameters of radon in porous material by flowthrough diffusion experiment, Appl. Radiat. Isot. 45 8 (1994) 845–850.
- [61] KELLER, G., HOFFMANN, B., FEIGENSPAN, T., Radon permeability and radon exhalation of building materials, Sci. Total Environ. 272 1–3 (2001) 85–89.
- [62] CHAUHAN, R.P., CHAKAVARTI, S.K., Radon diffusion through soil and fly ash: effect of compaction, Radiat. Meas. 35 (2002) 143–146.
- [63] ALDENKAMP, F.J., DE MEIJER, R.J., PUT, L.W., STOOP, P., An assessment of *in situ* radon exhalation measurements, and the relation between free and bound exhalation rates, Rad. Prot. Dosim. 45 1–4 (1992) 449–453.
- [64] HUTCHINSON, G.L., LIVINGSTON, G.P., HEALY, R.W., STRIEGL, R.G., Chamber measurement of surface–atmosphere trace gas exchange: Numerical evaluation of dependence on soil, interfacial layer, and source/sink properties, J. Geophys. Res. Atmos. 105 D7 (2000) 8865–8875.
- [65] LIVINGSTON, G.P., HUTCHINSON, G.L., SPARTALIAN, K., Trace gas emission in chambers: a non-steady-state diffusion model, Soil Sci. Soc. Am. J. 70 (2006) 1459–1469.
- [66] MAYYA, Y.S., Theory of radon exhalation into accumulators placed at the soil-atmosphere interface, Radiat. Prot. Dosimetry 111 3 (2004) 305–318.
- [67] SAHOO, B.K., MAYYA, Y.S., Two dimensional diffusion theory of trace gas buildup in soil chambers for flux measurements, Agric. Forest Meteorol. 150 (2010) 1211–1224.
- [68] ZHUO, W., IIDA, T., FURUKAWA, M., Modeling radon flux density from the Earth's Surface, J. Nucl. Sci. Technol. 43 4 (2006) 479–482.

- [69] MAYYA, Y.S., EAPPEN, K.P., NAMBI, K.S.V., Methodology for mixed field inhalation dosimetry in monazite areas using a twin-cup dosimeter with three track detectors, Radiat. Prot. Dosim. 77 (1998) 177–184.
- [70] NIKOLAEV, V.A., ILIĆ, R., Etched track radiometers in radon measurements: a review, Radiat. Meas. 30 (1999) 1–13.
- [71] FLEISCHER, R.L, et al., Dosimetry of environmental radon: methods and theory for low-dose, integrated measurements, Health Phys. 39 (1980) 957–962.
- [72] SCOTT, R.D., MACKENZIE, A.B., Measurement of low ²²²Rn activities by means of surface barrier detectors, Nucl. Instrum. Methods Phys. Res. A 238 (1985) 160–164.
- [73] HOWARD, A.J., JOHNSON, B.K., STRANGE, W.P., A high sensitivity detection system for radon in air, Nucl. Instrum. Methods Phys. Res. A 293 (1990) 589–595.
- [74] ZAHAROWSKI, W., WHITTLESTONE, S., A fast portable emanometer for field measurement of radon and thoron flux, Radiat. Prot. Dosimetry 67 (1996) 109–120.
- [75] EAPPEN, K.P., SAPRA, B.K., MAYYA, Y.S., A novel methodology for online measurement of thoron using Lucas scintillation cell, Nucl. Instrum. Methods Phys. Res. A 572 (2007) 922–925.
- [76] BIGU, J., ELLIOTT, J., An instrument for continuous measurement of ²²⁰Rn (and ²²²Rn) using delayed coincidence between ²²⁰Rn and ²¹⁶Po, Nucl. Instr. Meth. A **344** (1994) 415–425.
- [77] USMAN, S., SPITZ, H., LEE, S., Analysis of electret ion chamber radon detector response to ²²²Rn and interference from background gamma radiation, Health Phys. 76 (1999) 44–49.
- [78] GROSSI, C., VARGAS, A., ARNOLD, D., "Quality study of electret radon flux monitors by an 'in situ' intercomparison campaign in Spain", Sources and Measurements of Radon and Radon Progeny Applied to Climate and Air Quality Studies (Proc. Tech. Mtg, Vienna, 2009), IAEA, Vienna (2012) 39–48.
- [79] KOTRAPPA, P., DEMPSEY, J.C., RAMSEY, R.W., STIEFF, L.R., A practical EPERM (electret passive environmental radon monitor) system for indoor ²²²Rn measurement, Health Phys. 58 (1990) 461–467.
- [80] IIMOTO, T., AKASAKA, Y., KOIKE, Y., KOSAKO, T., Development of a technique for the measurement of the radon exhalation rate using an activated charcoal collector, J. Environ. Radioact. 99 (2008) 587–595.
- [81] LIVINGSTON, G.P., HUTCHINSON, G.L., SPARTALIAN, K., Diffusion theory improves chamber-based measurements of trace gas emissions, Geophys. Res. Lett. 32 (2005) L24817.
- [82] SCHERY, S.D., WHITTLESTONE, S., HART, K.P., HILL, S.E., The flux of radon and thoron from Australian soils, J. Geophys. Res. Atmos. 94 (1989) 8567–8576.
- [83] FERRY, C., RICHON, P., BENEITO, A., ROBE, M.C., Evaluation of the effect of a cover layer on radon exhalation from uranium mill tailings: transient radon flux analysis, J. Environ. Radioact. 63 (2002) 49–64.
- [84] JHA, S., KHAN, A.H., MISHRA, U.C., A study of the technologically modified sources of ²²²Rn and its environmental impact in an Indian U mineralised belt, J. Environ. Radioact. 53 (2001) 183–197.

- [85] SPEHR, W., JOHNSTON, A., The measurement of radon exhalation rates using activated charcoal, Radiat. Protect. Aust. 1 3 (1983) 113–116.
- [86] IIMOTO, T., TOKONAMI, S., MORISHITA, Y., KOSAKO, T., Application of activated charcoal radon collectors in high humidity environments, J. Environ. Radioact. 78 (2004) 69–76.
- [87] PAPASTEFANOU, C., An overview of instrumentation for measuring radon in soil gas and groundwaters, J. Environ. Radioact. 63 (2002) 271–283.
- [88] RUCKERBAUER, F., WINKLER, R., Radon concentration in soil gas: a comparison of methods, Appl. Radiat. Isot. 55 (2001) 273–280.
- [89] BUNZL, K., RUCKERBAUER, F., WINKLER, R., Temporal and small-scale spatial variability of ²²²Rn gas in a soil with a high gravel content, Sci. Total Environ. 220 (1998) 157–166.
- [90] WINKLER, R., RUCKERBAUER, F., BUNZL, K., Radon concentration in soil gas: a comparison of the variability resulting from different methods, spatial heterogeneity and seasonal fluctuations, Sci. Total Environ. **272** (2001) 273–282.
- [91] MATOLÍN, M., KOUDELOVÁ, P., Radon in soil gas investigation and data standardization at radon reference sites, Czech Republic, Radiat. Prot. Dosimetry 130 (2008) 52–55.
- [92] NEZNAL, M., MATOLÍN, M., JUST, G., TUREK, K., Short-term temporal variations of soil gas radon concentration and comparison of measurement techniques, Radiat. Prot. Dosim. 108 (2004) 55–63.
- [93] CHEN, J., RAHMAN, N.M., ABU ATIYA, I.A., Radon exhalation from building materials for decorative use, J. Environ. Radioact. 101 4 (2010) 317–322.
- [94] SAHOO, B.K., et al., Estimation of radon emanation factor in Indian building materials, Radiat. Meas. 42 (2007) 1422–1425.
- [95] PETROPOULOS, N.P., ANAGNOSTAKIS, M.J., SIMOPOULOS, S.E., Building materials radon exhalation rate: ERRICCA intercomparison exercise results, Sci. Total Environ. 272 1–3 (2001) 109–118.
- [96] PICOLO, J.L., Absolute measurement of radon 222 activity, Nucl. Instr. Methods Phys. Res. A 369 (1996) 452–457.
- [97] DERSCH, R., Primary and secondary measurements of ²²²Rn, Appl. Radiat. Isot. 60 (2004) 387–390.
- [98] NUCLEAR REGULATORY COMMISSION, Characterization of Uranium Tailings Cover Materials for Radon Flux Reduction, NUREG/CR1081, USNRC, Washington, DC (1980).
- [99] LAWRENCE, C.E., AKBER, R.A., BOLLHÖFER, A., MARTIN, P., Radon-222 exhalation from open ground on and around a uranium mine in the wet-dry tropics, J. Environ. Radioact. **100** (2009) 1–8.
- [100] VALKOVIC, V., Radioactivity in the Environment: Physicochemical Aspects and Applications, Elsevier Science, Amsterdam (2000).
- [101] NIELSON, K.K., ROGERS, V.C., "Surface water hydrology considerations in predicting radon releases from water-covered areas of uranium tailings ponds", Geotechnical and Geohydrological Aspects of Waste Management (Proc. Symp. Fort Collins, 1986), Balkema, Rotterdam (1986) 215–222.

- [102] ROWE, R.K., QUIGLEY, R.M., BRACHMAN, R.W.I., BOOKER, J.R., Barrier Systems for Waste Disposal Facilities, 2nd edn, Taylor & Francis, Boca Raton, FL (2004).
- [103] NATIONAL RESEARCH COUNCIL, Assessment of the Performance of Engineered Waste Containment Barriers, National Academies Press, Washington, DC (2007).
- [104] EVANS, K.G., LOCH, R.J., Using the RUSLE to identify factors controlling erosion rates of mine soils, Land Degrad. Dev. 7 (1996) 267–277.
- [105] GIROUD, J.P., BONAPORTE, R., Leakage through liners constructed with geomembranes — part I. Geomembrane liners, Geotext. Geomembranes 8 1 (1989) 27–67.
- [106] STARK, T.D., CHOI, H., Methane gas migration through geomembranes, Geosynth. Int. 12 2 (2005) 120–125.
- [107] ZORNBERG, J.G., LAFOUNTAIN, L., CALDWELL, J.A., Analysis and design of evapotranspirative cover for hazardous waste landfill, J. Geotech. Geoenviron. Eng. 129 5 (2003) 427–438.
- [108] FALA, O., MOLSON, J., AUBERTIN, M., BUSSIÈRE, B., Numerical modelling of flow and capillary barrier effects in unsaturated waste rock piles, Mine Water Environ. 24 4 (2005) 172–185.
- [109] WAUGH, W.J., SMITH, G.M., BERGMANTABBERT, D., METZLER, D.R., Evolution of cover systems for the uranium mill tailings remedial action project, USA, Mine Water Environ. 20 (2001) 190–197.
- [110] LEWIS, B.G., McDONELL, M.M., Radon transport through a cool-season grass, J. Environ. Radioact. 4 2 (1986) 123–132.
- [111] KOZAK, J.A., REEVES, H.W., LEWIS, B.A., Modeling radium and radon transport through soil and vegetation, J. Contam. Hydrol. 66 3–4 (2003) 179–200.
- [112] INTERNATIONAL ATOMIC ENERGY AGENCY, Programmes and Systems for Source and Environmental Radiation Monitoring, Safety Reports Series No. 64, IAEA, Vienna (2010).
- [113] BOLLHÖFER, A., STORM, J., MARTIN, P., TIMS, S., Geographic variability in radon exhalation at a rehabilitated uranium mine in the Northern Territory, Australia, Environ. Monit. Assess. 114 (2006) 313–330.
- [114] INTERNATIONAL COMMISSION ON RADIATION UNITS AND MEASUREMENTS (ICRU), Sampling for radionuclides in the environment, J. ICRU, 6 (2006).
- [115] SHUKLA, V.K., SARTENDEL, S.J., RAMACHANDRAN, T.V., Natural Radioactivity level in soil from high background radiation areas of Kerala, Radiat. Prot. Environ. 24 (2001) 437–439.
- [116] UNITED NATIONS, Sources and Effects of Ionizing Radiation (Report to the General Assembly), Scientific Committee on the Effects of Atomic Radiation (UNSCEAR), UN, New York (2000).
- [117] ANDERSEN, C.E., Numerical modelling of radon-222 entry into houses: an outline of techniques and results, Sci. Total Environ. 272 (2001) 33–42.
- [118] HART, K.P., Radon Exhalation from Uranium Tailings, PhD Thesis, Univ. New South Wales, Kensington, Australia (1986).

- [119] ROGERS, V.C., NIELSON, K.K., MERREL, G.B., Radon Generation, Adsorption, Absorption and Transport in Porous Media, Rep. DOE/ER/60641, US Department of Energy, Washington, DC (1989).
- [120] NATIONAL COUNCIL ON RADIATION PROTECTION AND MEASUREMENTS, Measurement of Radon and Radon Daughters in Air, NCRP Report No. 97, NCRP, Bethesda, MD (1988).
- [121] NUCLEAR REGULATORY COMMISSION, Radon Attenuation Handbook for Uranium Mill Tailings Cover Design, Rep. NUREG/CR3533, USNRC, Washington, DC (1984).

NOTATION

A	surface area of an opening (m ²)
a	chamber radius (m)
С	radon concentration (Bq/m ³)
D	radon diffusion coefficient (m^2/s)
D_{M}	molecular diffusion coefficient (m ² /s)
D_{M}^{M}	molecular diffusion coefficient in air (m^2/s)
D_{MW}	molecular diffusion coefficient in water (m^2/s)
$D_{a}^{M''}$	effective (bulk) diffusion coefficient (m^2/s)
D_{u}^{e}	diffusion coefficient
É	emanation coefficient
ε	counting efficiency of the system relative to the activity of radon
	adsorbed (counts \cdot s ⁻¹ \cdot Bq ⁻¹)
f	radon flux density (Bq \cdot m ⁻² \cdot s ⁻¹)
$f_{\rm r}$	local flux density representing the activity of radon crossing per unit
- p	pore area per unit time $(Bq \cdot m^{-2} \cdot s^{-1})$
F	total radon exhalation flux (Bq/s)
G	average ore grade
Н	height (m)
$J_{\rm m}$	mass exhalation rate $(Bq \cdot kg^{-1} \cdot h^{-1})$
K	water-air partition coefficient
L_r	radon diffusion length (m)
λ	decay constant (s^{-1})
λ	effective decay constant for ²²² Rn, which is the sum of the leak rate
-	(if existing) and the decay constant (s^{-1})
М	mass (kg)
т	moisture saturation fraction (vol./vol.)
N	net count rate, after background subtraction, obtained during the
	counting period $t_{\mathcal{C}}$ (counts/s)
п	porosity
n_{s}	soil porosity
Р	radon production rate $(Bq \cdot m^{-3} \cdot s^{-1})$
P_p	radon activity released into the unit volume of the pore space per unit
*	time (Bq \cdot m ⁻³ \cdot s ⁻¹)
R	radium activity concentration (Bq/kg)
ρ_{b}	bulk density (kg/m)
ρ_g	grain density (kg/m)
ρ_w	density of water (kg/m)
Т	absolute temperature (K)
t	measurement time (s)

- t_c counting period
- d_{d} delay period from the end of exposure to the beginning of the counting interval (s)
- t_e period of exposure of the charcoal in the canister (s)
- τ tortuosity factor
- τ_{e} time constant of radon buildup
- v flow rate (m³/s)
- *V* effective volume (m³)
- *w* dilution factor (kg of solid waste for kg of uranium ore processed)
- z depth or thickness of the material (m)
- θ_d moisture content on a dry weight basis (vol./vol.)

CONTRIBUTORS TO DRAFTING AND REVIEW

Hansen, S.	Queen's University, Canada
Ishimori, Y.	Japan Atomic Energy Agency, Japan
Lange, K.	Canadian Nuclear Safety Commission, Canada
Martin, P.	Australian Radiation Protection and Nuclear Safety Agency, Australia
Mayya, Y.S.	Bhabha Atomic Research Centre, India
Phaneuf, M.	International Atomic Energy Agency
Sahoo, N.K.	Bhabha Atomic Research Centre, India



Where to order IAEA publications

In the following countries IAEA publications may be purchased from the sources listed below, or from major local booksellers. Payment may be made in local currency or with UNESCO coupons.

AUSTRALIA

DA Information Services, 648 Whitehorse Road, MITCHAM 3132 Telephone: +61 3 9210 7777 • Fax: +61 3 9210 7788 Email: service@dadirect.com.au • Web site: http://www.dadirect.com.au

BELGIUM

Jean de Lannoy, avenue du Roi 202, B-1190 Brussels Telephone: +32 2 538 43 08 • Fax: +32 2 538 08 41 Email: jean.de.lannoy@infoboard.be • Web site: http://www.jean-de-lannoy.be

CANADA

Bernan Associates, 4501 Forbes Blvd, Suite 200, Lanham, MD 20706-4346, USA Telephone: 1-800-865-3457 • Fax: 1-800-865-3450 Email: customercare@bernan.com • Web site: http://www.bernan.com

Renouf Publishing Company Ltd., 1-5369 Canotek Rd., Ottawa, Ontario, K1J 9J3 Telephone: +613 745 2665 • Fax: +613 745 7660 Email: order.dept@renoufbooks.com • Web site: http://www.renoufbooks.com

CHINA

IAEA Publications in Chinese: China Nuclear Energy Industry Corporation, Translation Section, P.O. Box 2103, Beijing

CZECH REPUBLIC

Suweco CZ, S.R.O., Klecakova 347, 180 21 Praha 9 Telephone: +420 26603 5364 • Fax: +420 28482 1646 Email: nakup@suweco.cz • Web site: http://www.suweco.cz

FINLAND

Akateeminen Kirjakauppa, PO BOX 128 (Keskuskatu 1), FIN-00101 Helsinki Telephone: +358 9 121 41 • Fax: +358 9 121 4450 Email: akatilaus@akateeminen.com • Web site: http://www.akateeminen.com

FRANCE

Form-Edit, 5, rue Janssen, P.O. Box 25, F-75921 Paris Cedex 19 Telephone: +33 1 42 01 49 49 • Fax: +33 1 42 01 90 90 Email: formedit@formedit.fr • Web site: http://www. formedit.fr

Lavoisier SAS, 145 rue de Provigny, 94236 Cachan Cedex Telephone: + 33 1 47 40 67 02 • Fax +33 1 47 40 67 02 Email: romuald.verrier@lavoisier.fr • Web site: http://www.lavoisier.fr

GERMANY

UNO-Verlag, Vertriebs- und Verlags GmbH, Am Hofgarten 10, D-53113 Bonn Telephone: + 49 228 94 90 20 • Fax: +49 228 94 90 20 or +49 228 94 90 222 Email: bestellung@uno-verlag.de • Web site: http://www.uno-verlag.de

HUNGARY

Librotrade Ltd., Book Import, P.O. Box 126, H-1656 Budapest Telephone: +36 1 257 7777 • Fax: +36 1 257 7472 • Email: books@librotrade.hu

INDIA

Allied Publishers Group, 1st Floor, Dubash House, 15, J. N. Heredia Marg, Ballard Estate, Mumbai 400 001, Telephone: +91 22 22617926/27 • Fax: +91 22 22617928 Email: alliedpl@vsnl.com • Web site: http://www.alliedpublishers.com

Bookwell, 2/72, Nirankari Colony, Delhi 110009 Telephone: +91 11 23268786, +91 11 23257264 • Fax: +91 11 23281315 Email: bookwell@vsnl.net

ITALY

Libreria Scientifica Dott. Lucio di Biasio "AEIOU", Via Coronelli 6, I-20146 Milan Telephone: +39 02 48 95 45 52 or 48 95 45 62 • Fax: +39 02 48 95 45 48 Email: info@libreriaaeiou.eu • Website: www.libreriaaeiou.eu

JAPAN

Maruzen Company Ltd, 1-9-18, Kaigan, Minato-ku, Tokyo, 105-0022 Telephone: +81 3 6367 6079 • Fax: +81 3 6367 6207 Email: journal@maruzen.co.jp • Web site: http://www.maruzen.co.jp

REPUBLIC OF KOREA

KINS Inc., Information Business Dept. Samho Bldg. 2nd Floor, 275-1 Yang Jae-dong SeoCho-G, Seoul 137-130 Telephone: +02 589 1740 • Fax: +02 589 1746 • Web site: http://www.kins.re.kr

NETHERLANDS

De Lindeboom Internationale Publicaties B.V., M.A. de Ruyterstraat 20A, NL-7482 BZ Haaksbergen Telephone: +31 (0) 53 5740004 • Fax: +31 (0) 53 5729296 Email: books@delindeboom.com • Web site: http://www.delindeboom.com

Martinus Nijhoff International, Koraalrood 50, P.O. Box 1853, 2700 CZ Zoetermeer Telephone: +31 793 684 400 • Fax: +31 793 615 698 Email: info@nijhoff.nl • Web site: http://www.nijhoff.nl

Swets and Zeitlinger b.v., P.O. Box 830, 2160 SZ Lisse Telephone: +31 252 435 111 • Fax: +31 252 415 888 Email: infoho@swets.nl • Web site: http://www.swets.nl

NEW ZEALAND

DA Information Services, 648 Whitehorse Road, MITCHAM 3132, Australia Telephone: +61 3 9210 7777 • Fax: +61 3 9210 7788 Email: service@dadirect.com.au • Web site: http://www.dadirect.com.au

SLOVENIA

Cankarjeva Zalozba d.d., Kopitarjeva 2, SI-1512 Ljubljana Telephone: +386 1 432 31 44 • Fax: +386 1 230 14 35 Email: import.books@cankarjeva-z.si • Web site: http://www.cankarjeva-z.si/uvoz

SPAIN

Díaz de Santos, S.A., c/ Juan Bravo, 3A, E-28006 Madrid Telephone: +34 91 781 94 80 • Fax: +34 91 575 55 63 Email: compras@diazdesantos.es, carmela@diazdesantos.es, barcelona@diazdesantos.es, julio@diazdesantos.es Web site: http://www.diazdesantos.es

UNITED KINGDOM

The Stationery Office Ltd, International Sales Agency, PO Box 29, Norwich, NR3 1 GN Telephone (orders): +44 870 600 5552 • (enquiries): +44 207 873 8372 • Fax: +44 207 873 8203 Email (orders): book.orders@tso.co.uk • (enquiries): book.enquiries@tso.co.uk • Web site: http://www.tso.co.uk

On-line orders

DELTA Int. Book Wholesalers Ltd., 39 Alexandra Road, Addlestone, Surrey, KT15 2PQ Email: info@profbooks.com • Web site: http://www.profbooks.com

Books on the Environment Earthprint Ltd., P.O. Box 119, Stevenage SG1 4TP Telephone: +44 1438748111 • Fax: +44 1438748844 Email: orders@earthprint.com • Web site: http://www.earthprint.com

UNITED NATIONS

Dept. 1004, Room DC2-0853, First Avenue at 46th Street, New York, N.Y. 10017, USA (UN) Telephone: +800 253-9646 or +212 963-8302 • Fax: +212 963-3489 Email: publications@un.org • Web site: http://www.un.org

UNITED STATES OF AMERICA

Bernan Associates, 4501 Forbes Blvd., Suite 200, Lanham, MD 20706-4346 Telephone: 1-800-865-3457 • Fax: 1-800-865-3450 Email: customercare@bernan.com · Web site: http://www.bernan.com

Renouf Publishing Company Ltd., 812 Proctor Ave., Ogdensburg, NY, 13669 Telephone: +888 551 7470 (toll-free) • Fax: +888 568 8546 (toll-free) Email: order.dept@renoufbooks.com • Web site: http://www.renoufbooks.com

Orders and requests for information may also be addressed directly to:

Marketing and Sales Unit, International Atomic Energy Agency

Vienna International Centre, PO Box 100, 1400 Vienna, Austria Telephone: +43 1 2600 22529 (or 22530) • Fax: +43 1 2600 29302 Email: sales.publications@iaea.org • Web site: http://www.iaea.org/books

13-18101



GENERIC MODELS FOR USE IN ASSESSING THE IMPACT OF DISCHARGES OF RADIOACTIVE SUBSTANCES TO THE ENVIRONMENT

Safety Reports Series No. 19 STI/PUB/1103 (216 pp.; 2001)

ISBN: 92–0–100501–6

Price: €51.00

REGULATORY CONTROL OF RADIOACTIVE DISCHARGES TO THE ENVIRONMENT IAEA Safety Standards Series WS-G-2.3

STI/PUB/1088 (43 pp.; 2000) ISBN: 92–0–101000–1

Price: €14.50

This report provides a comprehensive overview of the prediction, measurement and monitoring of radon releases from NORM residues. including uranium mining and milling residues. It presents factors controlling radon emanation exhalation from residue materials. and describes repository cover characteristics and details methods for predicting radon exhalation flux, including models and their required input parameters and variables. It describes measurement methods for radon concentrations in soil gas and for radon exhalation from a surface. Radon monitoring programmes are also discussed. Finally, a case study of radon exhalation from a uranium residue pile at Jaduguda, India, is included and the appendices present the main mathematical development of radon diffusion equations.

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA ISBN 978–92–0–142610–9 ISSN 0074–1914